AN ANALYSIS OF THE EXPRESSED TRAINING NEEDS OF COUNTY AGRICULTURAL EXTENSION AGENTS IN KANSAS

by

NAZEEH ABDUL-HADI

B. S., Kansas State University Manhattan, Kansas, 1961

A MASTER'S THESIS

submitted in partial fulfillment of the

requirements for the degree

MASTER OF SCIENCE

Department of Education

KANSAS STATE UNIVERSITY Manhattan, Kansas

1963

Approved by:

a for Froiessor

LD 2688 TY 1963 A23 C.2 Document

TABLE OF CONTENTS

LIST	OF T	Walles	page
CHAP	TER		
	ı.	INTRODUCTION	1
		Purpose of the Study	3
		Hypotheses	4
		Development of the Questionnaire	6
		Collecting of Data	7
		Respondents in the Study	7
		Analysis of Data	8
		Significance of the Study	9
		Definition of Terms	10
	II.	REVIEW OF LITERATURE	13
		Introduction	13
		The Concept of Training	13
ı		General Information About Training	15
		Determining Training Needs	18
		The Importance of the Competency Areas	22
		Need for Extension Training	26
		Types of Training Needs	30
		Induction Training Needs	30
		In-service Training Needs	33
		Graduate Training Needs	37
		Summary	42

CHAPTER

SITUATIONAL PACTORS RELATIVE TO TRAINING OF KANSAS AGRICULTURAL EXTENSION AGENTS.	43
	44
	45
	47
	50
	53
Relationship of Job Perception to Graduate Training	55
Relationship of Job Perception to Tenure	56
Sumary	57
EXPRESSED TRAINING NEEDS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE NIME GENERAL AREAS OF COMPETENCY	60
Comparison of Training Needs of Kansas agricultural Extension Agents Among the Nine General Areas of Competency	62
Training Needs and the Importance of the Nine Areas	62
Training Needs Compared by Tenure	64
Training Needs and Job Perception Groups	67
Training Heeds and Graduate Training	69
Training Needs and In-service Training	71
Summary	73
EXPRESSED TRAINING NEEDS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE SPECIFIC AREAS OF COMPETENCY	74
Agricultural Subject Matter	75
Research and Svaluation	80
Effective Thinking	25
Program Planning and Development	89
	94
	99
	Tenure Undergraduate Major of Agents Graduate Training of Agents Job Ferception as a Factor in Training Relationship of Job Perception to Undergraduate Training Relationship of Job Perception to Graduate Training Relationship of Job Perception to Tenure Sugmary EXPRESSED TRAINING MEMOS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE NIME GENERAL AREAS OF COMPETENCY Comparison of Training Needs of Kansas agricultural Extension Agents Among the Nime General Areas of Competency Training Needs and the Importance of the Nime Areas Training Needs and Job Perception Groups Training Needs and Graduate Training Training Needs and In-service Training Sugmary EXPRESSED TRAINING NEEDS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE SPECIFIC AREAS OF COMPETENCY Research and Evaluation

	The Admonttant Procession	103
	Understanding footal systems	105
	auterator Organization and Adalahatmatica	113
	Description of the contract of	117
VI	DEBLET, CONTROLONG AND RESIDENCE TRUSCHESSONS ASSESSMENT	119
	dinational Pasters Adabtive to Irmining of Humana Agricultural Antenedan Agentu	121
	IMMITTURE PROGRAMMENT AND PROPERTY OF THE PROP	122
	Oredante Training of Agents	122
	lindergreducte Rejor of Agail-Bearers	122
	Job Ferception as a Factor in Trebring	122
	Relationship of Job Perception to Undergraduate	122
45	Distinguing of Job Persuphien to Graduate Training	123
	Mintionship of Toward to Job Perceptions	123
	Training Reads Identified by Ramses Agricultural Extension Agents in the Mins General Areas of Competendy	123
	Openicie Items in Sach of the Hine Areas of Competency in which Emess Agricultural Entension Agents Desire Gredute and In-Service Training	757
	Recommendations	128
	APPANAL MALE CARACTOR OF THE STREET OF THE STREET CONTRACTOR OF THE STR	130
	RISLICORAPHY	131
	APPMDIX.	134

LIST OF TABLES

Tabl		Page
l.	Potential and Actual Kansas Agricultural Extension Agent Respondents by Fosition	8
2.	Tenure of Kansas County Agriculturel Satension Agents, July 1, 1962.	44
3.	Undergreduate Najor of Kaness Agricultural Extension Agents, July 1, 1962	45
4.	Institutions at which Kansas Agricultural Extension Agents Obtained their Undergraduate Degrees, July 1, 1962	46
5.	Ransas Agricultural Extension Agents Holding Mester's or Bachelor's Degrees Listed by District, July 1, 1962	48
6.	Semester Hours of Graduate Training Completed by Kansas Agricultural Extension Agents, July 1, 1962	49
7.	Kansas Agricultural Extension Agents Perception of their Frofessional Role	52
8.	Relationship of Job Ferception to Undergraduate Field of Training of Kansas Agricultural Extension Agents	54
9.	Relationship of Job Perception of Kansas Agricultural Extension Agents to the Amount of Graduate Training Received by Agents as Expressed by Academic Degrees	56
10.	Relationship of Job Perception by Kansas Agricultural Extension Agents to Graduate Training Completed, Expressed in Secontar Hours of Graduate Work	56
11.	Relationship of Job Perceptions of Kansas Agricultural Extension Agents to their Temure in the Cooperative Extension Service	58
12.	Kaneas Agricultural Extension Agents Perception of their Training Needs in the Mine Areas of Competency Compared with their Perception of the Importance of those Areas	63
13.	Kansas Agricultural Extension Agents Perception of their Training Needs in the Nino Areas of Competency Compared by Temure Groups	65
14.	Kanses Agricultural Extension Agents Perception of their Training Needs in the Nine Areas of Competency by Job Perception Groups	68

Table		Fage
15.	Graduate Training Desired by Kansas Agricultural Extension Agents in the Mine Areas of Competency	70
16.	In-Service Training Desired by Hansas Agricultural Extension Agents in the Hime Areas of Competency	72
17.	Kansas Agricultural Extension Agents' Opinions of their Training Needs in the General Area of Agricultural Subject Matter	76
18.	Oraduate Training Desired by Kansas Agricultural Extension Agents in the General Areas of Agricultural Subject Hatter	78
19.	In-Service Training Desired by Kansas Agricultural Extension Agents in the General Arcss of Agricultural Subject Matter	79
20.	Kansas Agricultural Extension Agents' Opinions of their Training Heeds in the General Area of Research and Evaluation	81
21.	Graduate Training Desired by Kansas Agricultural Extension Agents in the General Area of Research and Evaluation	83
22.	In-Service Training Desired by Hansas Agricultural Extension Agents in the General Area of Research and Evaluation	84
23.	Mansac Agricultural Extension Agents' Opinion of their Training Needs in the Gazeral Area of Effective Thinking	36
24.	Graduate Training Desired by Hansas Agriculturel Extension Agents in the General Area of Effective Thinking	67
25.	In-Service Training Desired by Kansas Agricultural Extension Agents in the General Area of Effective Thinking	68
26.	Kansas Agricultural Extension Agents' Opinions of their Training Needs in the General Area of Program Flanning and Development	90
27.	Graduate Training Desired by Eanoas Agricultural Extension Agents in the General Area of Program Planning and Development	92
28.	In-Service Training Desired by Kanses Agricultural Extension Agents in the General Area of Program Planning and Development	93
29.	Rances Agricultural Extension Agents' Opinions of their Training Needs in the General Area of Human Development	95
30.	Graduate Training Desired by Kansas Agricultural Extension Agents in the General Area of Human Development	97

Table		Page
31.	In-Service Training Desired by Kansas Agricultural Extension Agents in the General Area of Numan Development	98
32.	Kansas Agricultural Extension Agents' Opinions of their Training Needs in the General Area of Gammanication	100
33.	Graduate Training Desired by Kanese Agricultural Extension Agents in the General Area of Communication	101
34.	In-Service Training Desired by Kansas Agricultural Extension Agents in the General Area of Communication	103
35.	Kaneas Agricultural Extension Agents' Opinions of their Training Needs in the General Area of Educational Process	104
36.	Graduate Training Desired by Kansas Agricultural Extension Agents in the General Area of the Educational Process	106
37.	In-Service Training Desired by Mansas Agricultural Extension Agents in the General Area of the Educational Process	107
38.	Kenses Agricultural Extension Agents' Opinions of their Training Keeds in the General Area of Understanding Social Systems	109
39.	Graduate Training Decired by Kansas Agricultural Extension Agents in the General Area of Understanding Social Systems	110
40.	In-Service Training Desired by Kanses Agricultural Extension Agents in the General Area of Understanding Social Systems	111
41.	Kansas Agricultural Extension Agents' Opinions of their Training Reeds in the General Ares of Extension Organization and Administration	113
42.	Greduate Training Desired by Kansas Agricultural Extension Agents in the General Area of Extension Organization and Administration	115
43.	In-Service Training Desired by Kansas Agricultural Extension Agents in the General Area of Extension Organization and Administration	116

CHAPTER I

INTRODUCTION

The Cooperative Extension Service is the educational arm of the United States Department of Agriculture, and it is a part of the Land Grant College of every state. This agency which was created by the Smith Lever Act in 1914 emphasizes additional training for the Cooperative Extension employees. The primary function of the Extension Agent can be found in the Smith Lever Act which etates:

That in order to aid in diffusing among the people of the United States useful and practical information on subjects relating to agriculture and home economics and to encourage the application of the eame. . . That Cooperative Extension work shall consist of the giving of practical demonstrations in agriculture and home economics to persons not attending or resident in said colleges in the several communities and imparting to such persons information on said subjects through field demonstrations, publications and otherwise.

The Act clearly points out that the primary function of the Cooperative Extension Service is education. This educational responsibility is for all the people of the United States. This responsibility is increased and emphasized by the rapid development of science and technology. These developmente necessitate learning quickly in order that adjustment can be made to any new or unexpected situation. Adjustmente must be made by the Extension worker as well as by the dientele.

Continuous professional improvement is very essential for the Extension

Agent for many reasons - among them are the development of the scope of

U. S. Department of Agriculture, Ruling Affecting Cooperative Extension Work in Agriculture and Home Economics, Bulletin No. 285 (Washington: U.S. Government Frinting Office, Nevised, January, 1946), p. 7.

activities, and the rising level of education of the clientele, as well as the advances in science and technology. All the above reasons indicate that the Extension Agent needs more training from day to day. The development of Extension programs has pointed up the need for training in all the areas of subject matter, program planning, human development, home management and many other fields.

The responsibility of the Sooperative Extension Service lies in previding educational opportunities for the public, and this requires that the
Extension Service maintain high professional competence, and performance
among its workers. To maintain high professional competence, it has become
very important to identify the areas in which Extension workers need training,
the fields in which they decire training, and the training needs of Extension
Agents at various stages of tenure. This training should include all areas;
pre-esrvice, induction, in-service and graduate training.

Extension work is a profession. Agents are required to have certain qualifications, and introductory experiences before being considered in full standing within the profession. Qualifications which the Agent should have are judged by reference to essential criteria of a profession. Professional performance as discussed by Yoder reflects:

- 1. A formal and somewhat standarized training.
- It presumes the regular formalised exchange of information and experience emong the workers.
- 3. It reflects continual research.
- 4. It implies a knowledge of, and familiarity with, a continuous growing field of literature.

- It is guided by an accepted rule of professional ethics and stronger sense of public responsibility.

This study was designed to provide additional information on the training needs of Kansas Agricultural Extension Agents. This study relates to the attitudes, knowledge, and skills of the Agricultural Extension Agent. This study is directed to three questions:

- 1. In what areae do Kansas Agricultural Agents possess competence?
- 2. What training do currently employed agricultural Extension Agente in Kanses feel they need in these areas in order they can meet the needs of the public?
- 3. What are the areas in which Kansas Agricultural Extension Agente are interested in including in their graduate work or in-service training?

Purpose of the Study

The major purpose of this study was to analyze the training needs of County Agricultural Extension Agents in Kansas as these needs were expressed by each Agent.

Specific objectives were developed to give more detail to the study.

The specific objectives were:

Dale Yoder, Personnel Management and Industrial Relations, (4th ed. New York: Prentice Hell, Inc., 1956), pp. 26-27.

- To describe some of the personal and situational factors relative to training of Kansas agricultural Extension Agents. Such dara to include:
 - a. Tenure (experience).
 - b. Academic degrees held by Agricultural Extension Agents.
 - c. Subjects emphasized in undergraduate degrees.
 - d. Amount of graduate training already completed toward graduate degrees.
 - d. The perceived educational role of Agricultural Extension Agents.
- 2. To determine the expressed training needs of County Agricultural Extension Agents in Kansas in relation to the mine general areas of competency identified by the National Task Force on In-Service Training.
- To identify specific items, within the nine competency areas, in which Kansas Agricultural Extension Agents felt the most need to include in graduate and in-service programs.
- 4. To determine if there are associations or relationship between training needs as expressed by the Agents and:
 - a. Tenure
 - b. Ferdaived educational role

Hypotheses

The hypotheses for this study were developed to give direction to the study and to serve as a guide in collecting, analyzing, and summarizing of the date. The writer used null hypotheses in this study. The reason for the use of the null hypotheses approach are given in the following statement:

This negativistic approach to acquiring knowledge about a universe by formulating null hypotheses and then rejecting them

on the basis of evidence seems almost the equivalent of setting up straw men merely to shoot them down. Yet in so doing, certain logical possibilities are eliminated and the range of remaining possibilities is narrowed. It is a cautious way of proceeding as are most scientific procedures.

The hypotheses were based on the literature review, and the objectives of this study.

The data in this study were taken from a total population, and have been used as the basis for accepting or rejecting the null hypothesis. The following null hypotheses were developed:

- There is no relationship between the amount of training needed by Kansas Agricultural Extension Agents in each of the nine areas of competency and the importance of that area.
- 2. There is no relationship between the ranking of the nine general areas of competency on the basis of the amount of training needed by Kansas Agricultural Extension Agents and tenure.
- 3. There is no relationship between the ranking of the nine areas of competency of training needed by Kansas Agricultural Extension Agents compared by job perception groups.
- 4. There is no relationship between the amount of training needed by Kaneas Agricultural Extension Agents in each of the mine areas of competency and the graduate training desired.
- 5. There is no relationship between the amount of training needed by the Kansas Agricultural Extension Agents in each of the nine areas of competency and the in-service training desired.

Margaret J. Hagood and Daniel O. Price. Statistics for Sociologists, ('2nd ed. rev., New York: Henry Helt and Co., 1952), p. 238.

Development of the Questionnaire

The author with the approval of his graduate committee, decided to use the majority of statements of the questionnaire designed by McCormick. A questionnaire contained a total of 127 statements. The basic areas comprising the questionnaire were the nine areas of competency identified by the National Task Force on Cooperative Extension In-Service Training.

McCormick found that a more detailed sub-division of the nine areas would be necessary, if the resulting data were to be meaningful to supervisors and administrators in developing training programs for County Extension Agents.²

Each of the nine areas was divided into four parts as follows:

- 1. Part one, consorms the importance of the nine areas to the effectiveness of the Agricultural Extension Agents in Kensas.

 Agents could indicate the degree of importance as "absolutely essential", "highly desirable", "would be helpful", or "not important". A four point scale was developed for this part and will be discussed later.
- 2. Part two, concerns the amount of training the Kansas Agricultural Extension Agents feel they need in the nine areas of competency. Agents could indicate their training needs as "feel I understand" as the point of no training needed, through "need some training", "need such training", and "need very such training", as the point

Robert W. McCormick, "An Analysis of Training Needs of County Extension Agents in Chio", Unpublished Fh.D. Thesis, National Agricultural Extension Center for Advanced Study, University of Misconsin, 1959.

² Ibid., pp. 209-226.

- of the highest amount of training needed. Another item "not applicable to my job" was added. A four point scale was developed for all items.
- 3. Part three, concerns the areas which Kanses Agriculturel Extension Agents would like to include in their graduate training program, when taking leave of absence.
- 4. Part four, concerns the areas which the Kanene Agricultural Extension
 Agents would like to include in their in-service training program.

 Additional questions were added to the questionnaire to help in the
 analysis of the situational data. The questionnaire was protested by fellow
 graduate students at Kanene State University as a mosne of obtaining
 suggestions.

Collecting of Data

The data for this study were collected by the writer by sailed questionnaires to the Agricultural Lotension Agents in Kansas during the period of June, 1962. The author gave detailed instructions for filling out the questionnaire. The first part of the questionnaire was devoted to necessary background information about the respondents. The remainder of the questionnaire was devoted to information relative to each of the nine areas identified by the National Task Force on Geoperative Extension In-Service Training.

Respondente in the Study

All Kansas Agricultural Extension Agents who were on the job July 1, 1962 were included in this study. There were 105 potential respondents to the questionnaire.

Table 1 shows the potential, the actual, end the percentage of the respondents. The data show that 93.3 per cent of the Agricultural Extension Agents responded to the questionnairs.

TABLE 1. Potential and actual Kansas Agricultural Extension Agent respondents by position.

Position	Fotential	Actual	Percentage of
	Respondents	Respondents	Fotential
Agricultural Extension Agent	105	98	93.3

The percentage of respondents in this study was low compared to the percentage of respondents in similar studies conducted in Arkansas¹ and Chic.²
The Agents in the Arkansas study responded 100 per cent, and in Chic the response was 98.9 per cent.

Analysis of Data

The questionnaires completed by Earses Agricultural Extension Agents were pre-coded and the data were punched on IBM cards on equipments owned by the Statistics Department at Kancas State University.

A total universe and descriptive statistical techniques were used in this study. Data were analyzed in the following chapters by means of mean weighted ecores, percentage distributions, rank order, and coefficients of

landel K. Price, "An Analysis of In-Service Training Needs Identified by County Extension Agents in Arkansas," (Unpublished Ph.D. dissertation, Batienal Agricultural Extension Centerfor Advanced Study, University of Wisconsin, 1960), p. 22.

²NeCormick, op. cit., p. 20.

rank correlation.

Agricultural Extension Agents were asked to check the items they would like to include in their in-service, or graduate training. Respondents were asked to check a descriptive term as a measure of training needed. The terms "feel I understand", need "some", "much, "very much" training were used. The author assigned scores of one, two, three, and four respectively for each of the above terms. An equal distance was assumed between each descriptive term in order that the items compared could be put in rank order for comparative purposes in the analysis of the data.

Significance of the Study

The data of this research study revealed that the study has immediate application to the Kansas Agricultural Extension Service. The data should be helpful to the Extension training committee, administration, and members of the resident steff in planning Extension training programs. It should be helpful to the resident steff of Eansas State University when advising students both graduate and undergraduate interested in Extension Service.

The findings of this research study have their first application to the Kansas Extension Service. In addition, they will contribute additional information on the Extension training all over the country. States which have similar training problems, and organizations similar to Kansas' should be able to apply some of the findings of this study.

In addition, the study emphasizes the importance of the nine areas identified by the National Task Force on Extension In-Service Training, and reinforce McCormick's conclusions that the items used in this questionnaire

McGarmick, op. cit.

can be useful to any state in making a study of its training needs.

Definition of Terms

The following definitions are used for certain terms in this study.

Agricultural Extension Agents: A term used to designate all County

Extension Agents who are responsible for the agricultural program in Hannes.

Personnent: Sefera to the Agricultural Extension Agents in Kaneas who responded to the questionnaire.

Tenure: Refers to total number of years service, or length of service of County Agricultural Extension Agents in the Kanesa Extension Service.

General areas of training: Refere to the nine areas of competency identified by the National Tack Porce on Competative Extension In-Service Training. These areas were used as the framework for this study. The nine areas of competency are:

- 1. Extension organization and edministration,
- 2. Numen development,
- 3. The educational process,
- 4. Understanding social systems.
- 5. Frogram development,
- . 6. Communication,
- ~ 7. Effective thinking,
 - E. Technical subject matter,
 - 9. Research and evaluation.

Specific areas of training: Refers to those sub-areas listed under each of the nine general areas of competency identified by the National Task Force on Comporative Axtension In-Service Training. "New to analyze the situation

in my county", "How to use specialiets in program development" are sub-areas of training under the general area of program planning and development.

Agent: Refers to Agricultural Extension Agents.

Descriptive terme: Refers to measures used to indicate the degree of training County Agricultural Extension Agente feel they need. The descriptive terme are "feel I understand", "need some training", "need much training", and "need very much training".

Job perception groupe: Refere to two groups of County Agricultural Extension Agente. The first group was composed of Agente who perceived their role as that of an educator. The second group was composed of Agents who perceived their role as organizers, or purveyore of subject matter information, or providere of service, or the non-educational role.

Induction training: Training program which start the first day the new Agent is on the job, and continues the first year of employment. This training program is designed to familiarize the new Agents with the Extension organization, teaching methods, communication, training in subject matter, and other fields related to Extension work.

In-Service training: Training program which an experienced Agent receives after his appointment to the Kansas Extension Service. This program may include training in any of the nine areas identified by the National Task Force on Cooperative Extension In-Service Training, such as subject matter, program planning, communication, Extension teaching methods, and other fields related to Extension work.

Graduate training: This term refers to all formal coursee which an Agricultural Extension Agent takes after receiving a bachelor's degree, or those formal courses which are taken at the graduate level.

Chapter II deals with review of literature.

Chapter III deals with the situational factors relative to training of Kanous Agricultural Extension Agents.

Chapter IV deals with the expressed training meeds of Kanasa Agricultural Extension Agents in the general areas of computency.

Chapter V deals with the expressed training needs of Kansas Agricultural Extension Agents in the specific areas of training.

A summary, conclusions, and recommendations are presented in Chapter VI.

CHAPTER II

REVIEW OF LITERATURE

Introduction

Extension Services in many of the etatee are attempting to build a cound training program in order to improve Extension Agents! abilities and skills in performing their respective jobs. Hany investigations and attempte have been made to identify the training needs of County Extension Agents. It is chapter includes a summary of the literature related to the above subject. Books, bulletins, theses, and other materials related to Extension, business and industry have been reviewed, and the facts have been recorded. Some of the methods used by the Extension Service and industry to determine training needs are much the same and have been included.

No attempt has been made to summarize all the research dons in the area of identification of training needs. The purpose has been to review research data that help to support and establish the situation for the investigation, and to support and substantiate the findings of this particular study.

The Concept of Training

The identification of training needs of the County Extension Agent is a very important step in establishing a training concept. If one trice to locate a definition for the word training in a dictionary, he will find different kinds of definitions. None of these definitions are broad enough to cover all aspects of training in business, industry, and Extension.

In a study mads by DePhillips and associates on employees' training

needs, the term training was used to mean:

That process which, under company auspices, seeks in a planned, co-ordinated and continuous manner to develop attitudes, which will maximize individual present and future efficiency and the effectiveness of the over all company operations.

Training involves a variety of activities which the County Extension Agent and the firm employee undertakes to improve their abilities, skills, understanding, attitudes and appreciation. Training will qualify the employee and better prepare him for his job. Industry suggests this aim for training:

To build continuously and systematically to the maximum degree and in proper proportions, that knowledge and those skills and attitudes which contribute to the welfare of the organization and the employee.²

U The Extension Service has the same idea as industry and business regarding training. The Mational Task Force on Cooperative Extension In-Service Training has the following to say about in-service training:

In-service training and development of employees is essential to efficient operation of the Cooperative Entansien Service and to the attainment of its program goals. The purpose of such training is to prepare the workers for the immediate job shead, for changes in problems and situations as they affect the work to be done, and to stimulate professional improvement. Training shall be designed to develop vigorous critical thinking and balanced action to meet current challenges.

The literature revealed a similarity in the meaning of training among

Frank A. DePhillips, William M. Berliner, and James J. Cribbin, Management of Training Programs. (Illinois: Home Wood, Michard D. Irwin, Inc., 1960, p. 6.

²Marl G. Plenty, et al, Training Employees and Managers. (New York: Ronald Press Company, 1948), p. 25.

³An In-service Training Program for Cooperative Extension Personnel. Recommendations of the National Task Force on Cooperative Extension Inservice Training, p. 29.

business, industry and the Cooperative Extension Service. Industry has done much research on training. Much of this research can be used by the Extension Service in solving some of its own training problems.

General Information About Training

Training is an important process which helps to improve the employees of any firm or industry. It is a continuous process carried on throughout the year. Administrators usually try to train their employees every time a new discovery or development is found. On the other hand, older employees feel that they need continuous training in order to hold their jobs and improve their skills. New Agents sust learn new skills, and knowledges, and adopt new attitudes in order to become an integral and effective part of their organisation.

The suther found that there is little opposition to training. Next literature, new and old, considered training as an important and essential process in the building up of a good staff. The effectiveness of any organization depends largely on the work-and action of its employees. Every employee has satisfactory and unsatisfactory performances on the job. If an organization wants to be successful, then, it should try to correct unsatisfactory situations. Unsatisfactory work experiences can be improved by training. Bellows states:

If new workers could be given a capsule of magic ingredient, causing them to increase their production by fifty per cent, this would be a worderous accomplishment. In the absence of such magic, training is one of the answers.

Every day new training challenges are developing, as a result of the

Reger M. Bellows, Psychology of Personnel in Business. (2nd ed., Englewood Cliffs, New Jersey: Frentice Hall, Inc., 1954), p. 304.

rapid development of ecience, technology, and invention of machines.

Covernment, industry, and business are concerned with day to day training.

A spekeman for the United States Department of Labor said:

The transformation we're now seeing will make the 16th Century Industrial Revolution look like a 'pink tea'. If this is true then management will require highly trained and sidiled people to carry on ever increasing and complex business and industrial operations. And these people must be trained if they are to keep pace with technological progress. Mechanisation and automation not only have tended to reduce the number of dull and repetive jobs but have also tended to eliminate the number of dull, repetive workers. The responsibility of management is to select and train the best possible corps of employees.

The Cooperative Extension Service is also concerned with training. The Joint Committee Report on Extension Programs, Folicies and Goals met in 1948 and discussed the importance of the Extension training program. The committee summed up this importance in the following:

especially designed to fit workers for the profession. The basis philosophy should be to have Extension workers as well trained as possible in broad fundamentals during their nedergraduate work, and to develop them into well qualified, technical persons by in-service training after they are employed. Extension has become of sufficient importance as a profession to justify special consideration in the land-grant college curricula. Then workers are first employed, industion training is essential. This should be followed throughout the workers careers by continuous in-service training. Opportunities for in-service training should be made available on official time.

Training employers is essential in industry, business and Extension.

Unrover, training to be effective must be based on the training needs of
the individual and their respective organization. Yoder indicates that when
building any training program, consideration should be given to the needs of

DePhillips, op. cit., p. 4.

²Joint Committee Report on Extension Programs, Policies and Goals, W.S.D.X. and Association of Land Grant Colleges and Universities, Washington, B.C. 1948, p. 42, Government Printing Office.

the employees, and their organization. Training progress should be used where there is an apparent need for them. He points out, however, that training will not oure all the sickness of any given organization.

Extension is not the only agency concerned with the training of employees.

Industry, business and management are putting much emphasis on personnal training. They consider training to be their responsibility. Todar found that 90 per cent of the companies who report their employees' activities put much emphasis on employee training. Company executives consider employee training as one of their primary responsibilities.²

Training is needed, but, the need varies from one employee to another.

Every worker has had different kinds of experiences, and consequently every workershas different needs. The duty of trainers is to satisfy those needs, if the organization is to be successful. Extension administrators believe that Extension training should be based on the needs of the worker in any given organization. In the Fifth National Administrative Workshop held at the University of Misconein, Extension administrators reported:

- 1. A job description is the basis for determining training needs.
- 2. The people performing the job must be involved in deciding the specific functions of the job and training needed.
- 3. A statewide committee to savise on training is suggested.
 "Across the board representation to decide upon over-all training needs". (4-H, agriculture, home economics, special county workers, specialists, resident staff, experiment station personnel).
- 4. Her workers should be considered when thinking of tradbing needs.

Moder, op. cit., p. 282.

²Ibid., p. 279.

- Fotential promotions in the county and to specialists and supervisory positions should be a part of the considerations on training needs.
- Agents should have an opportunity to express their needs for training.

Companise spend much money on training. They take into consideration what training will do for the company. They want to gain a maximum benefit from svery dollar spent on training. Executive usually consult their employees before they plan their training program, in order to determine employee assed and to use the right measures to fill those needs.

Industrial and business firms have conducted much research and have given much thought to the problem. Business and industrial personnel feel that training programs should be based on the identified needs. One writer states:

All industry training should be based on a determination of what ie needed, whether the need be for an induction program, supervisory training or executive development, and that doesn't mean haphazard guessing of training neede, such as is currently prevalent in many companies. Before management establishes a formal training program, it must set up a specific program of objectivee, define how the plan is to be carried out, who is to be trained and why.²

Determining Training Neede

Usually employee training is carried on with a point of reference.

It should be related to both the present and the future neede of any given organization. The question most often asked is: "What is the basic process

Report of Fifth National Administrative Workehop, Cooperative Extension Administration, National Agricultural Extension Center for Advanced Study, University of Wisconsin, Madieon, 1956, p. 53.

²Jamse C. Dunbar, "Survey Training Neede," Personnel, Vol. 25, No. 3 (November, 1948), p. 225.

in identifying training needs?" The basic process as identified by the United States Civil Service Commission in:

- 1. Determine that is required or expected on the jeb or in the situation.
- 2. Determine the degree to which these requirements or expectations are being met. If they are not being met, find out the rescens. To the extent that these reasons involve changes in knowledge, skill, attitudes, or behavior of employees (et all levele) you have a need which training sen probably help meet. You and the line officials consermed will have to judge whether training is the best way to meet it.

Before building any Extension training program, consideration should be given to the determination of training needs of the Extension worker. It is very important to determine training needs, because by doing so action can be taken toward building a training program based on those needs. Duncan states the importance of determining training needs and how to go about it by this statement:

The most basic and top priority problem facing extension training committees when considering action on an extension training program is determining training needs and building the training program around these needs. The first step is that of a thorough analysis of the need for training, by administrators and an agreement on general objectives of the training program and a setting of administrative policy that will guide the development of a long time coordinated training program. The next step is to designate a committee representing all phases of extension and the college to act within the framework of the administrative policy, to work out procedures for determining training needs and to recommend a program to be followed.

There are several schemes which can be followed by a committee in determining what problems should form the basis of the training program. Duncan suggested a scheme whereby the training committee composed of all phases of

Assessing and Reporting Needs and Progress, United States Civil Service Commission, Foreignel Rethods Series No. 3, December, 1961, p. 7.

²Janes A. Dancen, Training Cooperative Extension Workers, The Coordinated Approach, Cooperative Extension Service, College of Agriculture, University of Fisconsin (Medison, April, 1957, p. 33.)

Extension vorters set forth training needs as they see them. This method to be effective requires prior preparation by the committee members in order to determine Extension worker thinking and on-the-job problems. Other methods in determining Extension worker's training needs are the use of survey questionnaires and the use of either personal or group interviews. A carefully planned and thoroughly implemented need-inventory should form the basis of each state long-range training program.

Industry employs many methods in determining training needs. Some of these methods have been identified by various authors. Bellows, Estep and Scholl suggested the use of "Training Evaluation Check List." Bellows feels that "Job Analysis" is a good method. Fryer thinks that training needs can be identified by "Studying the Learning Behavior of Trainees". MacGhee suggests the use of "Employee Progress Records". Dumbar suggests the use of the "Survey Questionnaire".

Other methods of identifying training needs are used. Defhillips in his book "Management of Training Progress" discusses the following methods and thinks that these methods may prove useful in assessing training needs:

1. Job analysis and specification.

¹ Ibid., p. 33.

Zhoger M. Bellows, Frances M. Estep, and Charles M. Scholl, Jr., "A Tool for Analysis Training Heeds, The Evaluation Check Lists," Personnel, American Management Association EXIZ, 1953, pp. 412-417.

Regiewood Chiffs, New Jersey: Frontice-Hell, Inc., 1954, p. 304.

ADroglas H. Fryer, "Training With Special Reference to It's Evaluation," Perconnel Psychology, IV, IV, 1951, p. 19.

Fulliam McGehee, "Gutting Training Waste," Personnel Psychology, I, 1948, p. 331.

⁶Dunbar, op. cit., pp. 225-231.

- 2. Performance rutings.
- 3. Analysis of personnel records.
- A, Other company records, cost and production records, abtendance records, and labor turnover records.

In a manary of general methods of need determination, the United States Civil Service Commission suggests the following methods:

- 1. Interviews.
- 2. westionnaires.
- 3. Teste.
- 4. Job analysis and performance reviews.
- 5. Group problem analysis.
- 6. Necords and reports studies,2

Taxtension uses similar methods in determining the training needs of its workers. The Matitual Tack Force suggested three steps:

- 1. Locating areas of training need throught
 - a. Analysis of jobs.
 - b. Analysis of pragram emphasis.
- 2. Identifying the individual worker in need of training through:
 - n. Self-marveys of meeds and interests.
 - b. Deg-to-day observation by supervisor and specialists.
 - c. Tests,
 - d. Analysis of performance evaluations.
- 3. Determining priorities in training seed, 3

DePhillips, on. cit., pp. 273-286.

Passessing and Reporting Training Needs and Progress, op. cit., pp. 12-13.

^{3.} An In-service Training Progress, op. cit., p. 9.

Some of the methodo used by industry and Extension to identify training needs are similar. Most researchers in the field believe that it is very important that the training of employees should be based on the training needs of morkers.

The Importance of the Competency Areas

In the training of Agents, the most important step is to identify the areas of competency. Leagans expressed his ideas about this when he said:

What is known today about the tosching learning process all points to the necessity of identifying the competencies to be developed in the learners as a necessary condition for effective training. Without this how does one know, for example what kind of subject anther to use, what kind of teaching procedures and techniques of communication to use, or how such time and effort to allocate to the test. When one examines the nature of recognized professional group, it is discovered that a prime condition is clarity of competencies expected, and also, standards for these. For evidence one needs only to examine such professions as:

Rediction, law, formal education, and that of others like carpenters, plumbers, and air plane pilots. In the training of each of those groups, the competencies aimed at are clearly identified and there are usually at least minimum standards for competencies required for the professional practice.

Leaguns suggested oleven areas of competency. He states that these items are not intended to be all inclusive or of aqual importance. They represent a wide range of literature related to Extension work and his twenty years of study and experience in Extension. The competency areas are:

- 1. Enceledge and understanding of subject matter.
- 2. Understanding Extension and its educational role.
- 3. Skill in human relations.
- 4. Ability to plan.

²J. Paul Leagens, Competencies Seeded by Extension Workers, Report of Conference on Extension Training, Cornell University, April, 1957, p. U.

- 5. Ability to clarify objectives.
- 6. Ability to organize.
- 7. Communication skill.
- 8. Skill in relating principle to practice.
- 9. Skill in inquiry.
- 10. Evaluation ability.
- 11. Ability to provide learning experiences or to teach.

Leagans points out that there is a continuous need to further clarify the competencies needed by Extension workers. 1

There are nine areas of competency which are generally considered important for all Extension workers. Those areas were identified by the National Task Force on the Cooperative Extension In-service Training.

- 1. Understanding the Cooperative Extension Service.
- 2. Understanding human development.
- 3. Understanding program planning and development.
- 4. The educational process.
- 5. Social systems.
- 6. Communication.
- 7. Philosophy and values.
- 8. Technology.
- 9. Mesearch and evaluation.2

The above nine general areas were solected as the framework for this study. These same areas were the basis for studying training needs of Extension Agents in Chio³ and Arkansas.⁴

libid., pp. 10-19.

²An In-service Training Program, op. cit., pp. 13-15.

³wcCormick, op. cit., pp. 4-5.

⁴Price, op. cit., p. 13.

Agent training needs in these competency areas differ from one state to another. In the Chio Study the rank order of the nine areas on the basis of the training that the Chio Agents said they needed at the time of that research project was:

- 1. Program planning and development.
- 2. Effective thinking.
- 3. Communication.
- 4. Technical knowledge.
- 5. Human development.
- A. Research and evaluation.
- 7. The educational process.
- 8. Understanding social systems.
- 9. Extension organization and administration.

The rank order of the nine areas on the basis of how important the respondents felt it was for the Extension Agents to be trained in these areas was:

- 1. Technical knowledge.
- 2. Program planning and development.
- 3. Communication.
- 4. Extension organization and administration.
- 5. The educational process.
- 6. Effective thinking.
- 7. Human development.
- 8. Understanding social systems.
- 9. Research and evaluation.

lacCormick, op. cit., p. 90.

The three areas in which Agricultural Agents indicated they needed the most training were:

- 1. Program planning and development.
- 2. Human development.
- 3. Effective thinking.

In Arkanese the nine areas were the basis for studying training needs of the Extension Agent. Price conducted the study in Arkanese.

The rank order of the competency areas, based upon the percentage of Agents who indicated each area as "very important" as to agent effectiveness, was as follows:

- 1. Program planning and development.
- 2. Understanding human development.
- 3. Communication.
- 4. Effective thinking.
- 5. Understanding social systems.
- 6. The educational process (tie).
- 7. Extension organisation and administration (tie).
- 8. Research and evaluation.

The technical knowledge area was not included in the question on importance because logically its importance was assumed. Without knowledge of technical subject matter, the Agent could not teach.

The following rank order of the competency areas is based upon the Agent's livting of the amount of training needed in each area. This rank order implies areas where heavier emphasis may need to be given. It does not imply that those areas ranked near the bottom are not important nor that Agents do not need training in these areas. The rank order for the areas was:

- 1. Program planning and development.
- 2. Understanding human development.
- 3. Technical knowledge.
- 4. Communication.
- 5. Research and evaluation.
- A. Effective thinking.
- 7. The educational process.
- 6. Understanding social systems.
- 9. Extension organisation and administration.1

il

Need for Extension Training

Extension plays a major role by taking the new findings of research to the farmers and home makers, and interpreting it for them. Then, Extension continues to help people to help themselves through well-trained Extension workers.

It is well known that the Cooperative Extension Service places great responsibility on its County Agents. The responsibilities of the Extension Agent and the need for continuous professional improvement has been stated by Dr. Glenn Frank, former President of the University of Wisconsint

The future of America is in the hands of two men, the investigator and the interpreter. And we have an ample supply of investigators, but there is a shortage of readable and responsible interpreters, men who can effectively play mediator between specialists and layment.

Science cwes its effective ministry as much to the interpretative as to the creative mind. Farely do the genius for exploration and the genius for exposition meet in the same mind. The investigator advances knowledge, the interpreter advances progress.

Price, op. cit.

²R. K. Bliss, et al, The Spirit and Philosophy of Axtension Work, (Washington: Graduate School, United States Department of Agriculture and the Epsilon Sigma Phi, 1952), 6. 335.

Extension agents are interpreters of the research found by the investigators. Interpreters should have the training and the willingness to think and translate the research work into the language of the farmers and the homesskers. To do so, and to be able to teach, transmit and interpret the findings of research, an Agent should be up to date in many subject matter areas.

Literature shows that Extendion administrators and supervisors are seriously concerned about training needs of Extendion workers, both new Agents and those who have been employed for a period of time. Administrators think that both parties should be prepared well enough to face their work with enthusiasm and good will. The concern of the administrators and supervisors is expressed by Duncan:

The need for professional training for Cooperative Extension workers now on the job, those preparing for Extension careers at the undergraduate level, and the present and anticipated need for qualified county workers, demands critical analysis of the status of present day Geoperative Extension Education. The training problems most urgent are those of increasing the competence of those on the job in such skills as Extension program development, Extension teathing methods, public policy, technical subject matter in agriculturs and home economics, Extension evaluation and public relations. It is further necessary to prepare new personnel to replace approximately 15 per cent of the county workers pack year due to resignation, retirement and newly created-positions,

Extension Agents should be well prepared and well trained if they want to meet today's challenges. Developments in science, technology and mechanization require Agents to keep abrest of those developments if they intend to be successful in interpreting new research findings. Extension Agents shift their programs to meet the changing world. The 1958 "Scope Report" emphasized the need for Extension Agents to be currently well?

lpancan, op. cit., p. 2.

prepared to meet the challenges of change and demand. The report states:

One constant characteristic of Extension work has been the necessity to shift programs and methods to meet ever-changing conditions and demands. Extension workers have been acutely aware of this need from the beginning. The tempo of such changes has been accelerated dramatically during the past decade. Every evidence points to an even faster acceleration in the decade shead, I

This eras in which we are living is in a setate of change. This indicates that Agente need continuous professional improvement as well as continuous education for all people. Calivar states some of the general changes:

- 1. Our population is increasing.
- 2. The population is more mobile.
- 3. The population is growing older.
 - 4. Easy and swift communication is creating one world.
 - 5. We are working less.
 - 6. We are producing more.
 - 7. Our work is easier.
 - 8. The family and home are undergoing a radical change. 2

Some of the changes in agriculture during the past forty years are stated by Myere:

- 1. There has been greater use of mechanical power.
- There has been gradual increase in the size of operation that a farm family can handle. This resulted in fewer by larger family farms.
- 3. The trend toward specialization of farm production continues.

lpaul A. Miller, et al, The Cooperative Extension Service To-day-A Statement of Scope and Responsibility (East Lansing, Michigan: Michig

Ambrose Calivar, "Continuing Learning for a Changing World," Adult Leadership, Vol. VI, No. 3, (Sept. 1957), p. 73.

- 4. The change from self-sufficient to commercial agriculture has meant more buying and selling.
- 5. There is an increasing intensity of production.
- 6. Transportation and communication have come far.
- 7. The increased use of printed publication, as well as radio and television helps to improve the effectiveness of our Extension program.¹

The above mentioned changes which occurred during the last forty years are very important. It is necessary then, to help farmers keep in closer contact with the changing conditions in all fields related to agriculture. We can help farm families help themselves to keep in contact through better trained Extension workers. These Extension workers should be trained from year to year, because we cannot wholly depend upon the skills and knowledges in prior years to meet the demends of today and temorrow.

A continuous professional training program is very necessary for Extension Agents, who want to improve their abilities, skills and knowledge, so they can do a better and more useful job.

Training programs, then, are a necessity for the development of Extension Agents. The Kensas Extension Service provides training programs for their County Agents. It provides induction training to acquaint the beginners with the organization, the purposes, methods and procedures, and to acquaint the new Agent with the Central Office personnel. Another type of training which the Extension Service in Kensas provides is in-cervice training which tries to improve the competency of County Agents, in order to meet the changing times.

No. 1. Myere, "Meed of Changing Agriculture," Report of Conference on Extension Training, Cornell, 1957 (Westington: U. J. A., Federal Extension Dervice), pp. 4-6.

Types of Training Needs

The manner in which Agents should be trained will vary from one Agent to another. The most occase types of training are: induction, in-service and graduate training. All the above types are used at one time or another to develop the abilities and increase the knowledge and skills of the County Agricultural Agent. The three kinds of training are discussed below with the greatest emphasis placed on in-service and graduate training.

Induction Training Needs. The most critical period in the career of any employee is his first year on the job. The beginning County Extension Agent needs a great deal of training in order to orient himself to his new job. The work and action which the new employee will experience during his first year on the job will have an impact on his effectiveness during the coming years. Supervisors and administrators will formulate their opinions about new employees in the first year of employment.

Induction training, then, ie a very important process. Every newly appointed employee has a definite need for induction training. Duncan emphasized the pressing need for attention to induction training when he stated:

One of the high priority tasks in any State Extension Program is placing on the job, and providing carefully planned and effective training for new Extension workers. The significance of this problem is pointed up by the fact that approximately 15 per cent of the county workers of most states have been on the job less than one year. This is considered the critical period in an Extension worker's life, and most of them require rather intensive training and close personal supervision in order to become productive and adjusted Extension workers and attain a high degree of job security in the chortest length of time.

Training new workers is one of the most important and difficult problems facing the Cooperative Extension Service. Induction training to be successful should have the support and the backing of the entire Extension stait.

Duncan, op. oit., p. 24.

This support will help, to a large extent, in the development of induction training programs. The National Task Porce on Cooperative Extension Inservice Training set up some objectives for induction training which will help the new Agent.

- Cot a correct concept of the nature, purpose, and scope of the Cooperative Extension Service.
- Develop an understanding of his functions and his role as an employee of the Cooperative Extension Service.
- Obtain the knowledge and skills necessary to do his job and work in an efficient and effective manner.
- 4. Pevelop a "feeling of belonging" to an important educational organization.
- 5. Understand and appreciate the relationships of 4-41 work, home economics, and agriculture to a total Extension program.
- 6. Understand what constitutes a successful Extension Agent and how wall he meets the standard.

The induction training program starts the first day the new Agent is on the job and continues throughout the first year of employment. This program will supplement the training the new Agent had before starting on the job, and help the Agent put into practice what he learned during his preservice training. The training program should take into consideration the Agent's background and experience.

Many attempts have been made to find out what new experiences should be offered to the newly appointed Agents. Leagens has suggested the following experiences for new Agents during their first year on the job:

Injuction Training For County Extension Agents, Recommendations of the National Task Force on Cooperative Extension in-service Training, p. 2.

Memby appointed Agents should spend several days with different Agents in order to participate in the various phases of Extension work. Then, new Agents should have a conference at the district level. In these conferences, Agents will discuss the problem which face them, and to clear up questions that confront them in their job. They will learn what they can about the Extension Service and its various phases.

Agents should be encouraged to do some readings on the objectives of the Extension Service in relation to practical operation. This will help the new workers to become better acquainted with their job.

When planning any training program consideration should be given to the content of the program, the methods and the procedures used for training. In addition, consideration should be given to individual personal characteristics, college training, and the type of experiences new Agents have had. A comprehensive induction training program includes:

- 1. A brief orientation at the state office, under guidance of the supervisors.
- Field experience under direction of good trainer Agents in special training counties.
- 3. Visits to other counties to observe good Extension work.
- 4. Specific reading assignments.
- A new worker's conference after two to six menths' service, dealing with Extension organization, policies, philosophy and techniques.
- 6. Group training in subject matter and methods based on needs of individuals.

¹ Faul J. Leagans, "Suggestions for Induction Training of County Extension Servers", U.S.D.A. Kimoograph.

- 7. Pursonal conferences by supervisors to appealse the progress and plan specialized training needed.
- 2. There courses or special workshops on communication, organization and planning, and teaching techniques.
- A final evaluation conference with the supervisor at the end of the induction training to analyze the progress, give recognition and suggest additional improvement needed.

The review of literature indicated that when training newly appointed Agents, consideration should be given to the planning procedures to be followed in the training process. This usually differs from one Agent to another

In-service Training Needs. The job of the Geoperative Extension Service is educational. It is a continuous job in which new information, practices and new methods are developed; then transferred to the farmer for use. The job of the Extension Agent is to change attitudes, skills, knowledge, appreciation and the understanding of the farmers through education. Thus, County Agents should keep professionally equipped for their job. This is considered the most important responsibility of the Geoperative Extension Service. A downty Agent may be good enough on the job, but there is a continuous need for new information in the nine areas mentioned in this study and many other areas, in order that the Agent can do what is expected of him.

The Joint Committee Report on the Extension Programs, Policies, and Goals emphasize the need for continuous training for the Agents. They stated:

In-service training, of course, should not end with induction education. It should be a continuous process. It should be pleased to meet the needs of individuals and special groups, it should give opportunities for study under supervision, and it should be intensive. It is importent that Extension morkers have apportunities for frequent short periods of

Induction Training, op. cit., pp. 2-3.

in-corvice training to bridge gaps in sandomic proparation, to serve as refresher training, and to meet changing problems and situations as they affect the job to be done.

Collings in a poper proposed in 1954 brings out the facts that the current thinking of Geoperative Extension administrators and supervisors emphasing two major reasons for a strong in-service education (in-service training) program. These ares

- The resoval of deficiencies in pro-service and previous inworvice proposables and.
- 2. Continuouse of growth of Extension personnel.

To maintain the recognition of Cooperative Extension as a teaching profession, both State and Federal administrators are concerned with the need for professional improvement of workers.

Saveral studies have been made which encourage the in-service training program. However, most of those studies are broad and identify general areas of training pather than specific areas.

A study made in New York State showed that the Agricultural Agents placed less weight on training in a number of subject matter fields and more on therough knowledge in one field than Home Demonstration and 4-H Club Agents. This came obudy showed that the three areas of content preferred by the Agent for further training were in the following rank order: (1) improvement of techniques and skills on the job, (2) subject matter, and (3) social sciences.

Joint Committee Report, op. cit., p. 45.

²Mary L. Collings, "In-service Education of Cooperative Extension Workers." Federal Extension Service (Washington: U.S. Government Printing Office, 1954), p. L.

³harold W. Bonn and Charles R. Aiken, "A Look at Our Job. Factors Associated with Effective Extension Teaching in New York: (Ithaca, New York: Cornell University, 1953), p. 11 (Cited by McCormick, p. 54).

A Texas study revealed that downty Agents attached more importance to training in the areas of social science than any other area. Social science was considered to include such subjects as Scenomics, Socialogy, Education and Schemeion Methods.1

The in-service training of County Agents is accomplished by different means. These wary from individual training by supervisors from one to three days spent in a district, to a state wide conference, workshop or short course which lasts from one to three weeks.

Collings gives the following state in-corvice education programs, and their description, in a paper written in 1954. This list is becoming generally accepted in Extension training circles as a means through which in-corvice training is accomplished:

- 2. Summer school courses. These courses are initiated by States and Limited to their own personnel. Normally, they are of three weeks duration and serve as refresher causes in subject matter as well as methods. They provide for full-time attendance at college chases designed expecially for the needs of Autonaton workers.
- Committees to study specific problems. The purpose of such work is, in part, to provide extension workers with an opportunity for individual growth while working on common problems.
- 3. Interchange of work responsibility between Agents, A recent study of 4-H club work indicated that a considerable block of the Agent's time was spont in exchanging responsibilities with other Agents.
- 4. Clinics. News writing, radio speaking, and television work are so new that intension agants often have had no training in them during collage. Clinics, in which agents practice speaking, and showing, and then see and hear themselves perform, are popular and halpful.

Liosoph L. Matthews, "A Method for Determining the Training Need of County Agents as the Basis for Planning Training Progress". (Unpublished Ph.B. thesis, University of Chicago, 1950), p.58. (Cited by McCommick, p. 45.)

- 5. Studies. Mace its beginning, the Entension Service has used county surveys to determine the extent of adoption of recommended practices by farm people. These and other studies of Extension are outstandingly effective methods of im-service education.
- 6. Field tripe to experiment stations. Since the main function of Attension workers is to divine significant research fingings in agriculture and home scenarios, and to encourage their acceptance, it is imperative that they be kept in touch with current developments in existence. Regular and dystematic field trips to observe experiments under way at college and Federal stations in a part of the in-service education program in all States.
- 7. District and State conference. Annual conferences, like the field trip, are paragree among the eldest methods most generally used by States. District conferences on a more frequent schedule have substituted for the State conference in seas instances.
- C. Leader-training meetings. The Entencion specialist's training or itself isdeers is an excellent vehicle for agent training as well. In the home economics phases special attention has been given to the leader-training meeting as a way of expanding and improving automaton work.
- 9. Professional associations. County intension Agents have bended themselves into three professional organizations. The Matienal Association of County Agricultural Agents, the Matienal Association of Here Description Agents, and the Matienal Association of County Club Agents are organized opporately yet work cooperatively and session each other in sumerous ways. Bash association has a professional imprevement committee. These standing committees testify to the fact that through their our efforts Agents are seeking to raise the level of their work.

Literature shows that consideration has been given to the value of in-service training by Extension Agents in different states. Also, the Extension Services put much emphasis on this area of training - more than on any other area. Extension Services held training sessions every year in order to try to improve the efficiency of their employees.

It is apparent that County agents have had different kinds of

localings, op. ait., pp. 6-7.

experiences, and this means that in-service training needs are different from one Gounty Agent to another. The area of in-service training will require much more research in order to give more information about inservice training needs.

Graduate Training Needs. Graduate work in Extension Education for County Agents is becoming an essential part of every training program. The increase in job requirements, the complexity of the job, and the increasing number of Extension workers, all bring into focus the need for graduate work. Graduate study should be planned to improve the ability and skill of the Agent to do his job, or prepare him to do a better job.

Duncan in this treatise of training Geoperative Extension workers listed ten important factors giving rise to the need for graduate training in Extension Education. These factors were given by Leagans in a talk to the staff of the Federal Extension Service in 1956 and are as follows:

- Standards for professional proficiency are constantly rising in all fields.
- 2. Effective Extension work results from choice not from chance.
- It is an intricate and complex educational task today to design and execute Extension programs that significantly change the actions of large numbers of people.
- 4. Education is the central force in effective extension work.
- 5. The central idea in Extension work for "helping people learn how to help themselves" has proven to be a "good idea".
- 6. Effective educational leadership requires that a gap exists between what the leaders know and can do and the followers know and can do.
- 7. It is not what a person merely knows, but what he becomes to believe that determines what he does when he is free to act as he chooses.
- 8. To raise the level of living of farm people requires that Extension gives attention to:

(a) The family and its individual members.

(b) The home as a physical unit, (c) The farm as a business enterprise, and

- (d) The keeping of these in reasonable balance.
- 9. There are two major areas in which Extension workers must have adequate profession competence.

(a) Technical subject matter, or what to teach.

- (b) Educational process, or how to teach.
- 10. There is now developed a body of knowledge shout Extension education that is being recognized by University graduate schools as a major field of study leading to both master's and doctor's degrees.

From the foregoing points Leagans draws four conclusions which seem quite dsfsnsible:

- 1. No longer is four years of undergraduate training in any field enough for Extension workers. There is too much to be known, there is too much we must know to permit an end of formalized study after four years of undergraduate work.
- 2. In-service training programs, at their best, are not adequate for the job.
- 3. Something "new" is now created in some of our graduate schools designed to mest the special professional needs of Extension workers.
- 4. Adequate training for Extension work requires increasing attention to advanced study.

In ambitious Agent looks always to improve himself, in order that he might do s better job. One means of improvement is through graduate study. The Joint Committee Report in the following statement recommends graduate study as a manner of training most beneficial.

In recent years many graduate schools have attempted to arrange programs adapted to the need of Extension workers. Shortcourses of from thres to sight weeks have been provided. However, evidence points to the conclusion that graduate work for relatively short periods is not proportionally as beneficial as the more extended periods of study. Efforte should be made to provide leave

Duncan, op. cit., pp. 40-41.

to Extension workers for graduate study on quarter or semester basis.1

The most important responsibility of the Cooperative Extension

Services is to keep their County Agents professionally equipped. It has
been mentioned that each County Agent has different needs. The training
necessary to meet those needs must be provided primarily on the job.

Graduate work is becoming more important from year to year. This is due
to the development of the Extension Service and the increased complexity
of Extension work. Job requirements for the County Agent are increasing
and this points out the need for higher training, such as graduate study.

A 1957 study showed that four-fifths of the County Agents in Texas not only thought that graduate training was important but actually indicated that they would take leave from their job to do some graduate work, if some financial aid was provided.²

A study conducted in Chio by McCormick revealed that Extension Agents have been actively participating in graduate training programs. Two per cent of the Agricultural Agents have taken more than twenty hours of graduate work.3

A study made in Arkansas showed that 60 per cent of the Agricultural County Agents were interested in programs of graduate study toward a master's degree.4

The above studies show that there is a recognition of the need for

¹ Joint Committee Report, op. cit., p. 46.

²Benjamin D. Cook, "Comparative Analysis of Training Needs of County Agricultural Agents in Texas" (unpublished Ph.D. thesis, University of Wiaconain, 1957, p. 160.

³ McCormick, op. cit., p. 60.

⁴price, op. cit., p. 45.

graduate training among Extension workers. This need has been increased over the past years.

Many attempts have been made to identify the kind of training that County Extension Agents would like to include in their advanced study program. This is a very important area and should be taken into consideration when planning any graduate program.

A study of male Ohio and Wisconein Agents in 1949 showed the subjects that the Agents indicated they would take, if additional graduate training were taken. The following are the ten subjects, listed in rank order, that these Agents felt would be most holpful:

- 1. Extension education-psychology and principles of education.
- 2. Agriculture journalism technical news writing.
- 3. Speech-group thinking and conference leadership.
- 4. Organization of well and crop management systems,
- 5. Plant diseases control.
- 6. Advanced form management.
- 7. Extension evaluation.
- 8. Frinciples of adult education.
- 9. Planning rural communication educational programs.
- 10. Fural youth and social living.

Gook found that the majority of Agriculturel Agents who majored in technical agriculture as undergraduates expressed a desire to do their advanced study in technical agriculture. However, 45 per cent of these Agents majoring in technical agriculture as undergraduates expressed a

liscornick, op. cit., p. 49.

desire to do their advanced study in Extension Education or other Social Science fields. Other Agricultural Agents indicated that they were interested in taking graduate work in Extension, Sociology, and Psychology, if they were given opportunity to take such courses. Seventy per cent of all Agricultural Agents in this study, regardless of their undergraduate major, expressed the desire that they would like to do their graduate study in the Social Science area.

Harold Howe, Dean of Graduate School, Kansae Stats University and member of the Land-Grant College Committee on Pre-service and Graduate Training made a study of the graduats programs in Extension Education offered by Land-Grant Colleges. Hs stated that nine Land-Grant Inetitutione have programs leading to a master'e degree in this field. Two of the nine Institutions offer work leading to the degree of Doctor of Philosophy. Howe suggested that the program of study for any one student should be one of planned and integrated study in a chosen field of concentration. He suggested that a study leading to a master's degree in Extension Education consist of a major (approximately two-thirds of program) in a combination of Extension study, Communication and Social Science and a minor (approximately one-third of program) in technical subject matter. Howe suggested that ths thseis should be dons in the major field and specifically in the area of Extension study. Howe defines "Extension study" as including course work in Adult Education and in Extension Philosophy, Principles, Methods, Organization and Evaluation.2

From the literature review, the writer found considerable emphasie on

¹ Cook, op. cit., p. 154.

Duncan, op. cit., pp. 42-43.

graduate training. Leaguns explusives this need by saying that a four year degree is no longer enough for Extension workers. In-service training is not adequate for the job. Graduate schools are now offering special training to fulfill the needs of Extension workers. Administrators and supervisors should give attention to graduate study and encourage Extension Agents to participate in graduate training. In addition, opportunities should be provided to Extension Agents so they can do graduate work in the desired field.

Summary

The literature review shows that training needs of the worker should be determined before building any training program. The reason is that training needs are different from one Agent to another. Therefore, all Extension Agents cannot be given the came training and the same treatment. Appervisors are responsible for the determination of the training needs of County Extension Agents. They should assist and guide county workers in their training programs.

Training areas are many and the need for training in the different areas are very important. The remaining problem is how to identify the training needs of County Extension Agents. There are many methods for identifying training needs, but an effort should be made to develop aids and devices for the supervisors to use in identifying the training needs of County Extension Agents. This research study is directed toward this purpose.

CHAPTER III

SITUATIONAL FACTORS HELATIVE TO TRAINING OF EANSAS AGRICULTURAL EXTENSION AGENTS

It was indicated in the introduction of this paper that the purpose of this study was to analyse training needs of the Agricultural Extension Agents in Kansas. An analysis of the personal and situational factors relating to the respondents will help to clarify some of the findings of this research study. In addition, some of the findings may have application to other states which have situations similar to Kansas.

The author recognizes that the situations of the Kansas Agricultural Extension Agents will change from year to year, but the questionnaire patterned after McCormick's was used to determine the existing situations in Kansas in the year 1962.

The purpose of this chapter is to analyse the personal and situational factors relative to Ernsas Agricultural Extension Agents who responded to the questionnaire. The analysis includes:

- a. Termre or length of time employed.
- b. Degrees held by the Agricultural Extension Agents employed currently in Kansas.
- Rajor subject areas of agriculture emphasized in undergraduate major.
- d. Amount of graduate work completed by the Agents toward graduate degrees, and the interest of Agricultural Extension Agents in graduate study.

incomick, op. cit., pp. 209-226.

- e. Institutions from which Kansas Agricultural Amtenaton Agente obtained their undergraduate degrees.
- f. Agent's opinions of their role as educational or non-educational.

Tenasive

Table 2 shows the frequency distribution and percentages of Eansas Agricultural Extension Agents who responded to the questionnaire in the various tenure groups. Sumber of years of employment has been carried to the nearest year.

TABLE 2.—Termine of Kaness County Agricultural Extension Agents, July 1, 1962.

	County Agricu	ltural Agent
Tanure ^a	Munoer	Per Cant
Less than one year	1	1.0
1 - 5 years	32	32.7
6 - 10 years	33	33.7
11 - 15 years	36	16.3
16 - 20 years	7	7.1
21-2-25 years	4	4.1
26 - 30 years -	4	4.1
Over 30 years	1	1.0
Total.	98	100

To the mearest completed year.

The data show that 32.7 per cent of Kansas Agricultural Extension
Agents who responded to this study had less than six years of tenure. Thirty

three per cent of the Agricultural Agents had over two years service. Thirty three per cent of the Agents had 6-10 years service. Only one Agent had less than one year of service.

Undergraduate Major of Agents

All Agricultural Extension Agents employed by the Kansas Cooperative Extension Service have bachelor's degree in agriculture. While Kansas State University does not have an undergraduate major in Cooperative Extension work, there are courses in Extension Education such as Extension Organization and Policies and Extension Teaching Methods tought on the compus. These courses give an introduction to the Cooperative Extension work, to those students who are considering Extension as a career.

The highest per cent of Agricultural Extension agents majored in Anisal Eusbandry in their undergraduate work. Table 3 shows the frequency distribution and persentage of Kanses Agricultural Extension Agents in the different fields of study.

TABLE 3.—Undergraduate Major of Kansas Agricultural Extension Agents, July 1, 1962. N = 98

Undergraduate Major Field		ultural Agents
	Touber	For Cent
Agriculture Meananics	22	22
Agriculture Education	16	18
Agronomy	15	15
Animal Nucbandry	30	30
Dairy	5	5
Others	10	70
Total	100#	100

^{*}Two Agents have double majors.

The data show that 30 per cent of the Agricultural Extension Agents in Kansas who responded to this study majored in Animal Busbandry. Over one-fourth of the Agricultural Extension Agents majored in Agricultural Economics, eighteen per cent majored in Agricultural Education and fifteen per cent majored in Agronomy. Only five per cent of the Agricultural Extension Agents majored in Spairy, and 10 per cent majored in other fields,

Table 4 shows the frequency distribution and percentages of the Kansas Agricultural Extension Agents in this study who obtained their undergraduate degrees at various institutions.

TABLE 4.—Institutions at which Kansas Agricultural Extension Agents obtained their undergraduate degrees, July 1, 1962.

Institution	County Agric	cultural Agent
	Mulpar	fer Cent
laneas State University	73	73.7
Other Kansas Schools	1	1.0
Oklahoma State University	20	20.2
iniversity of Missouri .	2	2,0
Colorado State University	**	*
Iniversity of Nebraska	2	2.0
Other out of State School	1	1.0
Total	99*	100

Wone Agent has a degree from two different schools.

Many of the Kaneas Agricultural Extension Agents who responded to the questionnaire obtained their undergraduate training at institutions other than the Kaneas State University. This means that about 26 per cent of the Kaneas Agricultural Extension Agents have been exposed to curricula other than that of Kaneas State University.

Many of the Kaness Agricultural Extension Agents were graduates of Oklahoma State University. The data shows that about three-fourths of the Kaness Agricultural Extension Agents received their undergraduate training at the Kaness State University. One-fifth of the respondents received their undergraduate training at Oklahoma State University. The remainder of the respondents received their undergraduate training at the University of Missouri, the University of Mabrasha, other Kaness schools, and other out of state schools.

The above data show that Kanesa State University is the primary institution for undergraduate training of Kanesa Agricultural Astension Agents. Other institutions, primarily out of state, provide approximately 25 per cent of the undergraduate training for Kanesa Agricultural Extension igents.

Graduate Training of Agents

Erstante training is a vital part of the total training program of the Kanesa Extension Service. Agents interested in graduate training may apply for subbatical leave after six years of employment with Eurosa State University. Subbatical leaves are granted for study on a one year's basis at half pay or five months at full pay.

Table 5 shows the frequency distribution and the persentage of Eanses

Agricultural Agents who hold master's or bachelor's degrees listed by district. The distribution of Agents with master's degrees varied, with Southwest district having the highest percentage, and the Central district having the lowest percentage of Agents with master's degree. The data also show that 84 of the Agricultural Agents hold a beckelor's degree, and only 14 Agricultural Agents included in this study have sermed a master's degree.

TABLE 5.—Keness Agricultural Extension Agents holding member's or bechelor's degrees listed by district, July 1, 1962.

District	Number and P	erscent of Agents		Cent of Agents or's Degrees Col;
	Measer	Por Unas	Space r	l'er Cent
Northeast	3	21.4	16	78.6
Southeast	3	21.4	17	78.6
Central	1	7.1	20	92.9
Morthmeat	3	21.4	16	78.6
Southwest	4	26.6	15	72.4
Total	24	99.9*	E4,	

Trementages did not add to one landred, because a percentage table was

The date show that 14 per cent of the Kansas Agricultural Extension Agents included in this study have samed master's degrees. This per cent is considered high if compared with the nine per cent of the Agricultural Agents who earned master's degree in Arkansas', on the other hand the per cent is

Price, op. cit., p. 29.

considered low if compared with the 16.3 per cent of the Chic Agriculturel Extension Agents who earned master's degree, and the 17 per cent of the Agricultural Extension Agents in the United States who earned master's degrees.

Table 6 chose the semester hours of graduate training completed by Reneas Agricultural Extension Agents included in this study July 1, 1962. Semester hour is the term used at Kancas State University for expressing the amount of graduate training completed by Agricultural Extension Agents.

TABLE 6. -- Semester hours of graduate treining completed by Ennance Agricultural Extension Agents, July 1, 1962.
H = 98

lemester Hours of Graduate		ouldural agent
fraining Completed	Lumber	ler cent
lone	37	37.8
ine to aix	28	28.6
deven to trolve	13	13.3
Thirteen to eighteen	2	1.0
lineteen to twenty four	-	prode
Westy four to thirty	4	4.1
over thirty	15	15.3
Total.	98	100.14

^{*}Forcentages do not equal one hundred because a percentage table was used.

lucCornick, op. cit., p. 29.

Zourtis Trent, Forward Through Training, Cooperative Extension Service, Kansas State University, Manhattan, Mansas, May, 1762, No. 5, D. 2.

The data show that Kensus Agricultural Extension Agents have been participating in graduate training programs. Only 37.8 per cent of respondents have not participated in any graduate work. Participation of respondents in graduate training programs was not high. Only 19 per cent of the respondents have taken more than twenty-four hours of graduate training. Fifteen to sixteen hours of graduate study is approximately one-half of the requirements for a master's degree at Kansas State University.

The data show that 62 per cent of Kensus Agricultural Automaion Agents participated in graduate training while 14 per cent of the respondents have earned a master's degree.

Job Perception as a Faster in Training

One of the basic factors shown to be related to training in the Chic Intension Service was that of the perception of the Agents of their sole as an Agent. An understanding of what one's job is, what is expected of one by the people with whom he works in the county and what is expected of one by his supervisors in the Extension Service is a matter of prime importance. In the most important point of whether or not Agricultural Extension Agents in Emmas visual their professional role as that of an educator is discussed in this chapter. This point is considered very important and may have a bearing on the training needs expressed by the respondents.

In the Extension Service Service, the Federal Extension Service indicates that the role of the Extension Service is educational and the role of the

Macoratck, op. cit., p. 60.

Agent is education.

The following statement appears each month in the top of the first page:

The Extension Service Meview is for Extension educators - in Gounty, State and Federal Extension agencies - who work directly or indirectly to help people learn how to use the newest findings in agriculture and home economics research to bring about a more abundant life for themselves and their community.

Since the Extension Service Review indicated that the role of the County Extension Agent is education, a list of the five statements included in the questionneire of this study was used to determine the opinions of Kansas Agricultural Extension Agents as to their role. The following role statements were included:

- 1. A professional agriculturist available to provide information to the people in your county.
- A professional agriculturist providing service to the people of your county.
- A professional educator developing educational programs with people to effect behavior changes in the people in your county.
- 4. A professional educator developing programs to help people to help themselves.
- A professional organizer of educational activities for the people of your county.

Statements 1, 2, and 5 in the above liet were designed to show that the respondents to those items considered themselves non-educators. All Agents who selected the above three items have been considered playing the non-educational role.

¹ The Extension Service Neview, Vol. 31, No. 12, December 1960, p. 234.

Statements 3 and 4 in the above list were designed to show that the respondents to those items considered themselves educators. All Agents who selected the above two items have been considered playing the educational role.

Agricultural Extension Agents were asked to select the statement which best described the role they had been performing in their county previous to this study.

Table 7 shows how the Kansas Agricultural Extension Agents perceived their role as an Agricultural Agent. Thirteen per cent of Agricultural Extension Agents perceived their role as that of professional agriculturist, providing information. Six per cent perceived their role as a professional agriculturist providing service to people. Four per cent perceived their role as professional organisers of educational activities.

TABLE 7.—Kansas Agricultural Extension Agents perception of their professional role

N = 98

Role Perception of Agents		Extension Agent
	Mumber	Fer Cent
Professional agriculturist, home economist or youth worker providing information.	13	13.3
Professional agriculturist, home economist or youth worker providing service to people.	6	6.1
Professional educator developing educa- tional programs to effect behavior change in people.	46	46.9
Professional educator developing programs to help people help themselves.	29	29.6
Professional organizer of educational activities.	4	4.1
Total	98	100

Over 23 per cent of the Agricultural Extension Agents perceived their job as that of non-educators. This indicates that those Agents need more training to clarify the role they should play to help farmers.

These data show that over 76 per cent of the Kansas Agricultural Extension Agenta perceived their role as that of an educator. This is considered to be a high percentage of Agents who hold the same opinion as the Editor of the Extension Service Review. Over 23 per cent of the Agents perceived their role as that of non-educatora. The questions which arise here are: What group of Agricultural Extension Agents perform their job more effectively? Do the non-educator Agents perform as well as the educator? What are some of the factors associated with a non-educational perception of the job of an Agricultural Extension Agent? These areas are examined in the following pages.

Relationship of Job Perception to Undergraduate Training

This study has shown that a high percentage of Kansaa Agricultural Extension Agents majored in five different fields of Agriculture in their undergraduate training. Thirty per cent of the respondents majored in Animal Husbandry, 22 per cent majored in Agricultural Economics, 18 per cent majored in Agricultural Education, 15 per cent majored in Agronomy and five per cent majored in Dairy. The question here arises, does this undergraduate training have any relationship to job perception.

Table 8 shows the relationship of job perception to the undergraduate field of training of Kansas Agricultural Extension Agents. The data show that there was relatively little difference by undergraduate majors in the

TABLE 8.--Relationship of job perception of undergraduate field of training of Kansas Agricultural Extension Agents. 36 = 1

Job Ferception	-			Under	Lugar	ste Fie	Id of	Undergraduate Field of Training			Ì	
of the Agents	Noono Noono	Monthira Orosice Fer	Educa	tion	5 6	Fer	bandry No.	Anton fus- bandry No. Fer	3		2 3	ere for
		Cent		Cent		Cent Cent		Cent		Cent Cent		Cont
Education	ង	8	7	4.03	2	82.4 13 86.7 19		4 E.3	4	8	100	8
Won-educational	4	8	~	17.6	N	17.6 2 13.3	#	36-7	el	8	W	8
Total	8	100	11	100	33	15 100	30	100	W	8	2	201

percentage of Agents who perceived their job as educational. The Agents who majored in Agronomy and Agricultural Education had the highest percentage of those who perceived their job as an educator. The differences may not be important for the reason that all Agricultural Extension Agents in all tenure groups were included.

Relation of Job Perception to Graduate Training

This section of the study includes two parts. The first represents an analysis of the relationship of job perception of Eansas Agricultural Extension Agents to the amount of graduate training received by Agents as expressed in terms of scademic degrees. The second part is an analysis of the relationship of job perception to graduate training completed by Agricultural Extension Agents expressed in semester hours of graduate training. The division was made in terms of those Agents with less than minoteen samester hours of graduate training completed and those Agents with mineteen or more semester hours of graduate work completed.

Extension Agents with the amount of graduate training they had received as expressed by master's degrees. The data in this table show that a higher percentage of the Agricultural Extension Agent who held master's degrees perceived their role as educational in comparison with those Agents who held only a bachelor's degree. Minty-two point nine per cent of the Agricultural Extension Agents who held master's degree perceived their role as educational compared to 7.1 per cent who perceived their job as non-educational. Seventy-two point eight per cent of the Agricultural Extension Agents who held bachelor's degree perceived their role as educational compared with 27.2 per cent who perceived their job as non-educational.

TABLE 9.—Relationship of job perception of Eanses Agricultural Extension Agents to the amount of graduate training received by Agents as expressed by academic degrees.

949		district.
- 36	=	958

Job Ferception		Academic De		
of Agents		ris Degree		or Cent
	Munider	fer Cent	MATTO A.S.	701 0011
Educational Role	59	72.8	13	92.9
Non-educational Role	22	27.2	1	7.1
Total.	eı.	100	14	100

Three of the Agents did not respond to the job perception items on the questionmairs.

Table 10 shows the relationship of job perception of Kansas Agricultural Extension Agents compared with graduate training completed as supressed by the number of semester hours of graduate work completed.

TABLE 10.—Relationship of job perception of Kensea Agricultural Extension Agents to graduate training completed expressed in semester he rs of graduate work.

Semester Hours of Graduate Nork Completed Job Perceptions 19 hours and over O through 18 hours of Agents Fer Cent Manager P For Cent 88.9 16 72.7 Educational Pole 56 11.1 27.3 21 Mon-educational role 100 18 100 77 Total

Three of the Agents did not respond to the job perception items on the questionnaire.

The data in this table show that a higher percentage of Kensas Agricultural Extension Agents with over 19 hours of graduate work completed view their role as an educational one more than those Agents who completed less than 19 hours of graduate work. The data exphasize that people with 19 hours or more of graduate training tend to see their role as that of an educator more than those with less than 18 hours of graduate work.

Relationship of Job Perception to Tenure

An analysis of the relationship of tenure to job perception of Kansas Agricultural Extension Agents were made in this study. Tenure was essectiated only slightly with the Agent's perception of his job. Tenure may be considered a rough measure of the in-service training an Agricultural Extension Agent has received.

Table 11 shows the relationship of job perception of Kansae Agricultural Extension Agents to their tenure in the Cooperative Extension Service. The data in this table shows that the middle group of 6 to 20 years tenure include the highest percentage of the Kansas Agricultural Extension Agents who perceived their role as educational. The same group has the lowest percentage of Agents who perceived their role as non-educational. The two groups with 21 years and over, and 0 to 5 years of tenure included 75 per cent of the Agricultural Extension Agents who perceived their role as educational.

TABLE 11.—Relationship of job perceptions of Kansas Agricultural Extension Agents to their tenure in the Cooperative Extension Service.

N \pm 95⁸

Job Perception of	A-1-			Agents in		
Agents	0 00	5 Years		O Years	ZI lear	s and Over
	No.	Fer- Cent	No.	Per Cent	No.	Per Cent
Educational Role	24	75.0	42	76.4	6	75.0
Non-sducstional role	8	25.0	13	23.6	2	25.0
Total	32	100	55	100.	8	100

Three of the agents did not respond to the job perception item on the questionnaire.

Summary

The following is a brief summary of the results of the analysis of the data presented in this chapter:

- Kansas State University is the primary institution for the undergraduate training of Kansas Agricultural Extension Agents.
- Kansas Agricultural Extension Agents are participating in graduate training programs. Fourteen per cent of the Agricultural Agents have samed a master's degree.
- Thirty per cent of the Kansas Agricultural Extension Agents majored in Animal Husbandry. This was the highest percentage compared with other agricultural fields.
- 4. Approximately 76 per cent of the Kansas Agricultural Extension
 Agents perceived their role as that of educators, while 24 per
 cent perceived their role as that of non-educators.

- Graduate training and a master's degree increase slightly the probability of an Agent perceiving his role as that of an educator.
- Agents with 6 to 20 years of temme perceived their role as that
 of an educator to a slightly higher degree than any other temme
 group.

CHAPTER IV

EXPRESSED TRAINING NEEDS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE NINE GENERAL AREAS OF COMPETENCY

This chapter is devoted to the analysis of the expressed training needs of Kansas Agricultural Extension Agents in the general areas of competency identified by the National Task Force on Cooperative Extension In-Service Training. The main objective of this chapter is to determine the importance of the nine general areas of competency as reflected by Kansas Agricultural Extension Agents opinions of the importance of those items to Agent's success in their job performance.

The findings in this chapter will be helpful to Kansas Extension

Supervisors, Resident staff and Administrators. However, the methods used in identifying the training needs of Kansas Agricultural Extension Agents, may have implications for Extension Services throughout the country.

The data were analyzed for Kansas Agricultural Extension Agents who responded to the questionnaire. The data were taken from the responses of 98 Kansas Agricultural Extension Agents. Descriptive statist'cs were used in the study. Mean weighted scores, rank order, percentage distribution and rank order correlations have been used throughout the chapter. The procedure used in arriving at the mean weighted score was:

1. Values were assigned to each of the categories on the original questionnaire (See appendix). A four value was assigned to the "need very much training" category, a three value was assigned to the "need much training", a two value was assigned to the "need some training" category, and a one value was assigned to the "feel I understand" category.

- 2. Equal distances between the various categories were assumed.
- The number of Agents responding in each category to each item on the scals was multiplied by the assigned value of that category.
- 4. The product was divided by the total-number of respondents to the four categories to which values were assigned.
- 5. If an Agent failed to respond to an item or indicated that the item was "no applicable to my job" the Agent was not included in the total number of Agents responding to the item.

In the scale dealing with the importance of the nine general areas, a value of one was assigned to the "not important" category, a value of two to the "would be helpful" category, a value of three to the "highly desirable" category, and a value of four to the "absolutely essential" category. The same procedures were used in determining the mean weighted score for each item.

(rho), a measure of rank order correlation, has been used as a descriptive statistic in this chapter and the succeeding one to show the degree of relationship or association between rank orders. The author used the Spearman rank correlation coefficient formula to compute the correlations. The formula is:

$$rs = 1 - 6 \times di^2$$

To compute rs. make a list of N subjects. Next to each subject's entry his rank for X variables and his rank for Y variables. Determine then the various values of di = the difference between the two ranks. Square each di, and then sum all values of di to obtain di². Then enter this value

and the value of N (the number of subjects) directly to the above formula.

Hypotheses are stated at the beginning of each area. After computing the data, rank order correlations were computed in order to accept or reject the null hypotheses. The author accepted the hypotheses when the correlation was below the level .600 and rejected it when the correlation was above .600.

Comparison of Training Needs of Kansas Agricultural Extension Agents Among the Mine General Areas of Competency

Training Needs and the Importance of the Nine Areas:

Hypothesis 1: There is no relationship between the amount of training needed by Kansas Agricultural Extension Agents in each of the nine areas of competency and the Agent's perception of the importance of that area.

Table 12 shows Kansas Agricultural Extension Agents' perception of their training needs in the nine areas of competency compared with their perception of the importance of those areas. The Agricultural Extension Agents ranked the following as the top three areas in which they needed training:

- 1. Technical knowledge.
- 2. Research.
- 3. Effective thinking.

Sidney Siegel, Non-paramatric Statistics for the Behavioral Science. New York: McGraw Hill Book Co., Inc. 1956, p. 206.

TABLE 12. Keness agricultural Extension agents perception of their training needs in the nine areas of competency compared with their perception of the importance of these areas.

S . 98

Areas of Competency		seds Compared	d Source of Agent's with Importance	
	Reighted Score	Pank	weighted Score	and y
Technical knowledge	2.40	1	3.51	1
Research	2,22	2	2.87	5
Effective thinking	2.18	3	2.72	7
Progrem pleaning and development	1.98	4	2.91	4
Human development	1.94	5	2.53	8
Communication	1.91	6	3.24	3
The educational process	1.36	7	2,91	6
Secial systems	1.50	E	2.45	9
Extension organization and administration	1.66	9	3.26	2

The following areas were considered by the Kansas Agricultural Extension Agents to be the three areas in which it was most important for Agents to be trained, in order to be successful in performing their job:

- 1. Technical knowledge.
- 2. Extension organization and administration.
- 3. Communication.

The area of "Extension organization and administration" was rated last on the basis of need for training, while the area of "social systems" was rated last in terms of importance by Kansas Agricultural Extension Agents. The tesk erder correlation between training needed by "gricultural Extension Agents and the importance of those areas to Agents' effectiveness was: .216. This correlation is actual because it represents the whole population. This correlation is low.

The hypothesis was accepted. Receive the Gegree of relationship was very low this would indicate that there was no important relationship between expressed training needs and importance of the area to Agents' effectiveness.

The above results need not be too surprising, because the Agricultural Extension Agents may have had some training in many of the areas which they consider important in performing their job.

Training Roods Compared by Tenure

Hypothesis 2: There is no relationship between the ranking of the mine general areas of competency on the basis of the amount of training needed by Kansas Agricultural Extension Agents and tenure.

Table 13 shows the opinions of Kansas Agricultural Extension Agents relative to their training needs in the nine general areas of competency compared by tecure groups,

Agricultural Extension *gents with less than one to ten years temmer ranked the three general areas of training in which they felt the greatest need for training as follows:

- 1. Technical knowledge.
- 2. Effective thinking.
- 3. Research.

TAMES 12.—"Souses Agricultural "Attachen Agente perception of their training needs in the nine attens of competency compared by tennive groups.

If # 98

Areas of Coupesency	155 **	te and a	Sales of the sales	S S S S S S S S S S S S S S S S S S S	ARREST OR STREET TOTAL BY ARREST LEVING AND LIVE TO LIVE LIVE TO LIVE TO LIVE	CVE
	Seore	Park	Seom Seom	Frank	Section Contract	Pan J
Teshnical Enouledge	2,42	rt	2.39	7	2.50	н
Effective thinking	2.2	N	1.95	~	1001	101
Research	2.14	m	2,30	01	2	25
Harm development	200	mil ⁹	7.65	2	3.73	49
Fregren plenning and development	2,00	W	8	W	2,12	m
The educational process	2.98	•	1.73	1 40	7.50	93
Social cystems	1.69	2.5	1.63	43	1.50	123
Committelon	1.89	7.5	1.91	4	2.00	-4
Extension organization and administration	1.0	•	3.47	6	1.50	43

Agricultural Entension Agents having from eleven to twenty years of termre listed the following areas as the three most important in terms of their training needs:

- 1. Technical knowledge.
- 2. Research.
- 3. Offective thinking.

Agricultural Entennion Agents in the twenty one and over years of tenure group felt that the following three areas were the areas in which Agents need most training:

- 1. Technical knowledge.
- 2. Research.
- 3. Program planning and development.

The following rank order correlations were computed between the training needs of Agents in the various tempe groups:

Lose than one year to ten years and eleven to twenty years .804

Eleven to twenty years and twenty one years and over .863

Lose than one year to ten years and twenty one years and over .704

The above rank order correlations show that the lowest correlation was between the less than one year to ten years and twenty one years and over. The correlations in all three areas were high.

The hypothesis was rejected. The data presented suggests that training needs of Agricultural Extension Agents vary comewhat among different tenure groups, and this difference, though alight, may be an important point to consider when planning training programs.

Training Reads and Job Perception Groups.

Hypothesic 3: There is no relationship in the ranking of the aims areas of training needed by Esness Agricultural Automaton Agents compared by job perception groups.

Table 14 shows the opinions of Kansas Agricultural Extension Agents relative to the perception of the training needs in the nine general areas of competency compared by job perception groups.

Agricultural Extension Agents who perceived their job as scheational folt that the following areas were the most important in terms of their training needs.

- 1. Technical knowledge.
- 2. Effective thinking.
- 3. Recentch.

Agricultural Extension Agents who perceived their role as noneducational indicated the greatest intensity of need for training in the following areas:

- 1. Technical knowledge.
- 2. Besearch.
- 3. Effective thinking.

An analysis of the data in Table 14 showed, however, three areas of competency where little disagreement occurred between the sgents who perceived their role as educational and those who perceived their role as non-aducational.

The area of "Extension organization and administration" was rated last by Agricultural Extension Agents who perceived their role as educational and by Agents who perceived their role as non-educational.

TARLE 14. Fences Agricultural Artendern Agents perception of their training needs in the nine areas of emapetency by job perception groups.

N = 98

armon of Competency	sdacation M	the contribution of contribution and the financial contribution of the	res by Job Forceptio Non-Amonti N: 2	onal Pole
	Score	Pank	Score	Rank
Technical knowledge	2,40	1	2,41	1
Affective thinking	2,23	2	2.13	3
Rosesroh	2.19	3	2.26	2
Human development	2.07	4	1.80	8
Frogram planning and development	2.01	5	1.95	4.5
The educational precess	1.94	6	1.82	7
Commission	1,66	7	1.95	4.
Social systems	1.76	ŧ	1.84	6
Extension organization and administration	3.64	9	1.68	9

The rank order correlations of the training needs between Keneus Agricultural Extension Agents who perceived their job as educational and those who perceived their job as non-educational was .754. This was a high correlation between the ranking of training needs of the Agents who perceived their role as educational and those Agents who perceived their role as non-educational.

The hypothesis was rejected. The relationship was not strong but it suggests that training needs of Agents vary enaschet among different perception groups. This is an important point to consider when planning training programs.

Training Reeds and Graduate Training.

Hypothesis 4: There is no relationship between the amount of training needed by the Kansas Agricultural Extension Agents in each of the nine areas of competency and the graduate training desired.

Table 15 indicates the Agricultural Extension Agents' ranking of the nine general areas of competency in terms of graduate training the Agents desire to take in each area when he takes his leave of absence.

Rach Agricultural Extension Agent had been asked to indicate no more than three sub-items in each of the nine general areas of competency in which he desired to take graduate training when his next leave of absence for advanced study from the Kansas Extension Service became available. This meant that the greatest number of responses under each of the nine general areas of competency was 96 x 3 or 294. The actual number of responses under each of the nine general areas of competency is listed in Table 15. The percentage figure is based upon the percentage of the actual responses. The three areas that received the greatest number of responses in terms of graduate training desired in the nine general areas of competency were:

- 1. Effective thinking.
- 2. Human development.
- 3. Social systems.

The implications from the following data show that Kansas Agricultural Extension Agents do not desire to take graduate training, to a great extent, in the nine general areas of competency in which they felt the greatest need for training. Advisors for graduate programs should be aware of the training needs of the Agricultural Extension Agents as indicated on the

questionnaire used for this study, so they can provide the most useful graduate training program. Other implications to this study are: Supervisors should meet with Agricultural Extension Agents to find out what kind of training Agents need as they prepare to take their leaves of absence.

TABLE 15. Graduate training desired by Maneas Agricultural Extension Agents in the nine areas of competency.

Areas of Competency	Fositive F Training D	esponses of Agents on estred N = 294	Graduate
	humber	Per vent of Potential	ilank
Effective thinking	164	55.8	1
Numan development	94	32.0	2
Social systems	90	30.6	3
Program planning and development	88	29.9	4
Extension organisation and administration	73	24.8	5+
Technical knowledge	73	24.6	5•5
Communication	67	20.7	7
The educational process	57	19.4	8.
Research	57	19.4	€.

The rank order correlation between the /gente' ranking of the nine areas of compatency on the basis of the training moods in those areas and the ranking of the nine areas on the basis of the graduatetraining Agents desired to take when they take their leave of absence was .052.

The hypothesis was accepted. The degree of relationship was very low and this indicates that there was no important relationship between the training needs of Agents and graduate training desired.

Training Seeds and In-Service Training.

Hypothemis 5: There is no relationship between the amount of training needed by Kansas Agricultural Extension Agents in each of the nine general areas of competency and the inservice training desired.

Table 16 chows the rank order of the nine general areas of competency by Kansas Agricultural Extension Agents in terms of the in-service training Agents desired to include in their district next year. The procedures used in finding this ranking of areas were the same procedures used in Table 15.

The following were the most important three areas which Kansas Agricultural Extension Agents desire to include in their in-service training in the district next year:

- 1. Effective thinking.
- 2. Extension organization and administration.
- 3. Human development.

The implications from the above data indicate that Kansas Agricultural Extension Agents desire to participate in in-service training to a greater degree than graduate training. (Compare Table 15 with Table 16) Another implication is that there is a need for Supervisors to council with Agents in order that they may become aware of their in-service training needs as they prepare to participate in training programs.

TABLE 16. In-cervice training desired by Kansas Agricultural Extension Agents in the nine areas of competency. R=90

Area of Competency	Positive Training		ses of Agents d N = 98	on In-	-Servic
View or combacantel	Number		Fer Cent Potential		Finnk
Effective thinking	216		73+5		1
Extension organization and edministration	1.75		59.5		2
Human development	154		52.4		3
Communication	148		50.3		4
Research	130		44.2		5
Technical knowledge	122		41.5		6
Frogram planning and development	121		41.2		7
Social systems	113		38.4		8
The educational process	79	100%	26.9	de-	9

The rank order correlation between the Agents' ranking of the nine areas of competency on the besis of the training Agricultural Axtension Agents felt they needed and their ranking of the areas on the basis of the in-service training they desired to include in their in-service training program at the district level was .100. This descriptive statistics was very low, and this indicates that there was very little correlation between the above areas.

The hypothesis was accepted. The degree of relationship was very low and indicates no important relationship between the expressed training needs of Agricultural Agents and in-cervice training desired.

Daniel 177

The following were the major findings revealed by an analysis of the data presented in this chapter.

- Manuar Agricultural Extension Agents felt they needed the most training in:
 - a. Technical knowledge.
 - b. Research.
 - c. Effective thinking.
- Training needs of Agricultural Agents seemed to wary somewhat between different tenure groups.
- Whether or not an Agent perceived his job as that of an educator had little effect upon his ranking of the mine areas of competency on the basis of the training needed.
- 4. Kansas Agricultural Extension Agents do not desire to take graduate training nor desire to take in-service training on the district levelate any great extent in the areas of competency in which they expressed the greatest need.
- 5. There was a relatively low degree of relationship between the general areas of competency that Ransas Agricultural Extension Agents felt were important and the general areas in which Agents felt they needed training.

CHAPTER V

EXPRESSED TRAINING NEEDS OF KANSAS AGRICULTURAL EXTENSION AGENTS IN THE SPECIFIC AREAS OF COMPETENCY

Chapter four outlines Kaneae Agricultural Extension Agents' opinions as to their training needs in the nine general areas of competency identified by the National Task Force on Cooperative In-Service Training. Chapter four does not give detailed information for the epecific training needs within each of the nine general areas of competency.

This chapter presents an analysis of the expressed training needs of Kansae Agricultural Extension Agente in the specific areas of training listed under the nine general areas of competency identified by the National Task Force on Cooperative In-Service Training.

The data in this chapter were taken from the responses of 98 Kansas Agricultural Extension Agents. Descriptive etatistics have been used throughout the chapter. Mean weighted scores, rank order, percentage distribution and rank order correlation were used. The procedures used to compute weighted scores and rank order correlation in chapter four were used in this chapter.

Each of the nine general areas of competency was analyzed to determine the specific training needs of Agricultural Extension Agente. The analysis of the data in this chapter involved the Kansas Agricultural Extension Agents' ranking of the specific areas listed under each of the general competency areas. The ranking was based on three things:

- 1. Expressed training needs of Kansas Agricultural Extension Agents.
- Graduate training which Kansae Agricultural Extension Agents desire to take when they take their next leave of absence.

- 3. In-zervice training which Kensun agricultural Extension Agents
 desire to take during next year's in-service training program.
 Spearman's formulal, used in chapter four to compute the rank order
 correlations, has been used in this chapter to compute the rank order
 correlations between:
 - The specific training needs of Eansas Agricultural Extension Agents within each general area of competency and the graduate training desired by the Agents.
 - 2. The specific training needs of Kanens Agricultural Extension Agents within each general area of competency and the in-service training desired for next year's district training programs.

The analysis of the specific areas of training needs within the nine general areas of competency has been discussed in the same order as the Agent ranking of the nine areas of competency.

Agricultural Subject Matter

Table 17 shows the rank order and the weighted scores of the specific training needs of Kansas Agricultural Extension Agents within the general area of "Agricultural subject matter". The following were the three specific areas in "Agricultural subject matter" in which Agricultural Extension Agents indicated the greatest need for training:

 Agricultural Marketing (includes livestock, grain, dairy and poultry).

¹ Siegel, op. cit., p. 206.

TABLE 17. Kansas Agricultural Extension Agents' opinions of their training needs in the general area of Agricultural subject matter.

Specific area of Training	Scores of Age	
	seighted score	Mank
igricultural Marketing. (include livestock, grain, dairy and poultry).	2.50	1
igricultural Credit	2.41	2
Plant Pathology	2.39	3
Farm Management	2.37	4.5
igricultural Anginosring. (includes machinery, buildings and drainage).	2.37	4·5
Anisal Science	2.33	6
Interology	2.30	7
Norticulture. (includes penelogy and vegetable crops).	2,27	8
gronomy. (includes crops, soil, fertilizers)	2.23	9
Toriculture	2.19	10
Consumer Marketing. (market information for consumer)	2.03	11
Dairy Technology	1.99	12
Dairy Science	1.97	13
oultry	1.79	14
ioil Conservation	1.63	15

- 2. Agricultural Credit.
- 3. Plant Pathology.

Table 15 shows the Agricultural subject matter" which they desire to include in their graduate training. The following were the most important three specific areas of "Agricultural subject matter" which Agricultural Extension Agents desire to include in their graduate program when they take their leaves of absence:

- 1. Animal Science.
- 2. Farm Management.
- Agricultural Marketing (includes livestock, grain, dairy and poultry).

Table 19 shows the Agricultural Extension Agents' ranking of the specific areas in "Agricultural subject matter" that they desire to include in next year's in-cervice training program. The following were the most important three specific areas which Agents indicated that they desire to include in their in-cervice training program in their districts

- l. Animal Science.
- 2. Agroncay (includes crops, soil, fertilizers).
- Agricultural Marketing (includes livesbook, grain, dairy and poultry).

The rank order correlation between the training needs desired by Kanese Agricultural Extension Agents and the graduate training they desire to take in this area when they take their leave of absence was .746. The rank order correlation between the training needs desired and the inservice training Agents desire to include in their district in-service training program was .776.

TABLE 18. Greduate training desired by Eannes Agricultural Extension Agents in the general area of Agricultural subject matter.

Specific Ares of Training		Deales for Graduat	o Training
Animal Science	32	32.7	1
Vam Management	27	27.6	2
Agriculturel Marketing. (includes livestock, grain, dairy and poultry)	25	25.5	3
Agronomy (includes crops, soils, fertilisers).	24	24.5	4
Agricultural Credit	20	20.4	5
Horticulture. (includes penology and vegetable erops).	8	6,2	6
agricultural Engineering. (includes machinary, buildings and drainage).	6	6.1	7.5
Entanology	6	6.1	7-5
Dairy Science	5	5.1	9.5
Flant Fathology	5	5.1	9.5
Planiculture	3	3.1	11
Concurer Marketing. (Market information for concuser).	2	2,0	12
Soil Geneerwation	1	1.0	13
Dairy Technology	0	0.0	14.5
Poultry	0	0.0	14.5

TABLE 19.—In-service training desired by Kansas Agricultural Extension Agents in the general area of Agricultural subject matter.

Specific area of Training	Agents' D	eaire for In-Ser	rvice Train
	Musber	Fer Cent	Hank
inimal Science	33	33.7	1
igronomy. (includes crops, soil, fertilizers)	29	29.6	2
igricultural Marketing. (includes livestock, grain, dairy and poultry)	24	24.5	3
Form Management	23	23.5	4
Agricultural Gradit	22	22.4	5
echinery, buildings and drainage)	16	18,4	6.5
Flant Pathology	18	18.4	6.5
intenalogy	1.4	14.3	8
Ploriculture	11	11,2	9
Norticulture, (includes penalogy and vegetable crops)	10	10.2	10
Consumer Marketing. (Market infor- mation for consumer)	6	6.1	11
Dairy Science	5	5.1	12
Soil Conservation	2	2.0	13
Poultry	1	1.0	3.4
Dairy Technology	0	0	15

The rank order correlation between training needs and graduate training desired and training needs and in-service training desired in the district level, were moderately high and about the same for both semperisons. This data indicate that Agricultural Extension Agents desire in-service training, and graduate training in a sammer that will meet their training needs in the area of "Agricultural subject matter".

Research and Evaluation

Table 20 shows the opinions of Kenses Agricultural Actionsion Agents on their training needs in the general area of "research and evaluation". The following were the most important three specific areas in which Agents felt the greatest need for training within this general area:

- Enouledge of where to secure sound research information other than from experiment stations.
- 2. Understanding ways of designing evaluation projects.
- 3. Enouledge of the methods for measuring results of the programs.
- Making application of research findings, in assisting people in my country. (tie for third).
- 3. Now to make the best use of research publications. (the for third). Table 21 shows the Kanses Agricultural Extension Agents' runking of the specific area within "research and evaluation" which Agents desire to include in their graduate training program. The following were the most important three specific areas Agents desire to pursue when they take their next leave of absence:
 - Eaking application of research findings in assisting people in my county.

TABLE 20. - Kanses Agricultural Extension Agents' opinions of their training needs in the general area of research and evaluation.

Specific Ares of Training	Pank Order at Sucres of	
	selgated Score	Fenk
Inowledge of where to secure sound research in- formation other than from experiment stations.	2,13	1
Understanding ways of decigning evaluation.	2.10	2
Knowledge of the methods for measuring results of the programs.	2.05	4
making application of research findings in assisting people in ay county.	2.05	4
low to make the best use of research publications.	2.05	4
Inderstanding how to interpret statistical results.	2.01.	6
Now to interpret the results of research in the physical coiences.	1.95	7
ion to interpret the results of research in the social sciences,	1.93	8.5
Inderstanding the role of investigation in solving problems with people.	1.93	6.5
low to use the experimental approach (pilot project) in Extension work.	1.88	10
dow to develop simple survey devices.	1.67	23.
Inderstanding research terminology.	1.60	12

- 2. Encolding of the methods for mensuring results of progress.
- 3. How to use the experimental approach (pilot projects) in Madension work.
- 3. Understanding how to interpret statistical results. (tie for third)
- 3. Her to develop simple survey devices. (tie for third)

Table 22 shows the Kansas Agricultural Extension Agents' renking of the appendic areas within "research and evaluation" which they desire to include in their in-service training progress. The following were the three most important specific areas that Agents Indicated they desire to include in their district in-service training progress.

- 1. How to make the best use of research publications.
- Enouloge of where to meture wound research information other than from experiment stations.
- 3. Making application of research findings in assisting people in my county.

The rank order correlation between the training needs expressed by agricultural Extension Agents and the graduate training they desire to take in this area was .360. The correlation between the expressed training needs and the in-service training Agricultural Agents desire to include in their district next year was .812.

The relationship between the training needs of Agricultural Agents and the in-service training Agents desire to take was much higher than the correlation between training needs and the graduate training Agricultural Extension Agents desire to include in their graduate study program.

The data indicate that Kaness Agricultural Extension Agents will more nearly meet their expressed training needs through the in-service programs they desire to take in their district than through the graduate training they desire to include in their graduate "etudy program when they take their

Table 21.—Graduate training desired by Kansas Agricultural Extension Agents in the gameral area of research and evaluation.

Specific Area of Training	Agenta	Per Cent	7raining
isiding application of research finding in assisting people in my county.	s g	6.2	1
Enomicige of the methods for measuring results of programs.	7	7.1	2
How to use the experimental approach (pilot projects) in Extension work.	6	6.2	4
Understanding how to interpret statistical results.	6	6.1	4
Now to develop mimple survey devices.	6	6.1	4
Knowledge of where to secure cound research information other than from experiment stations.	5	5.1	6
Understanding ways of designing evaluation projects.	4	4.1	7.5
How to make the bast use of research publications.	4	4.1	7.5
How to interpret the results of research in the physical sciences.	3	3.1	1.0
How to interpret the results of research in the social sciences.	3	3.1	10
Understanding the role of inverti- gation in solving problems with people.	3	2.1	10
Understanding research terminology.	2	2,0	12

Table 22. -In-service training desired by Kaneas Agricultural Extension Agents in the general area of research and evaluation.

Specific Area of Training	Agents	Desire for In-		
Proposition constraints and automation of	Lomber	Per vent	Farik	
low to make the best use of research cublications.	23	23.5	1	
inculedge of where to secure sound research information other than from experiment stations.	21.	2.4	2	
Making application of research findings in assisting people in my country.	19	19.4	3	
Understanding ways of designing evaluntion projects.	12	12,2	4	
Understanding how to interpret stabletical results.	9	9.2	5	
insuledge of the methods for mossuring results of progress.	É	8.2	6.5	
Understanding the role of investigation in solving problems with people	e	€.2	6.5	
Now to use the experimental approach (pilot projects) in Extension work.	7	7.1	8.5	
How to develop simple survay devices.	7	7.1	8.5	
now to interpret the results of research in the physical extences.	6	6.1	10	
Now to interpret the results of research in the social sciences.	5	5-2	11.5	
Understanding research terminology.	5	5.1	11.5	

leaves of threace.

Effective Thinking

Table 23 shows the weighted scores and the rank orders of training dealered by Kanesa Agricultural Extension Agents within the general area of "affective thinking". The following were the three specific areas in which Agents expressed the greatest need for training within the general area of "effective thinking".

- Mnowledge of techniques for developing affective thinking in Extension groups.
- Understanding the role of the Extension worker in predicting probable future results from existing facts.
- 3. How to use the problem solving approach in Extension.

Table 24 shows the Kansas Agricultural Extension Agents' ranking of the specific training needs within the general area of "effective thinking" that they desire to include in their graduate study. The following three specific areas were the areas which Agricultural Extension Agents indicated the greatest desire to study when they take their next leaves of absence.

- 1. Understanding the processes of logical reasoning.
- 2. Understanding the role of the Extension workers in predicting probable future results from existing facts.
- Understanding the relationship of the "problem solving" method to effective thinking.

Table 25 shows the Eanses Agriculturel Extension Agents' ranking of the specific training within the general area of "effective thinking" which they desire to include in their in-service training program in their district next year. The following were the three specific areas which Agents desire:

TABLE 23.—Mannes agricultural Extension Agents' opinion of their training needs in the general area of effective thinking.

pecific Area of Training		nd Weighted f Agents
	esignted Sors	Basis
nowledge of techniques for developing feative thinking in Automaton groups.	2,22	1
derstanding the role of the Extension rivers in predicting probable future suits from existing facts.	2.19	2
on to use the problem colving approach a Antonoica.	2,10	3
nowledge of the principles of creative	2,04	Ą.
nderstanding the relationship of the problem solving" method to effective minking.	2.03	5
derstanding the effect of prejudice the thinking process.	1.494	6
derstanding the effect of pressure	1,93	7
derstanding the relationship of W-dreaming to creative thinking.	1.91	e
nderstanding the relation of language terms used) to the thinking process.	1,68	9
meretanding the processes of logical	1.63	10

Table 24. - discipate training desired by Kanase Agricultural Astencion Agents in the general area of effective thinking.

Specific Area of Training		Decire for Graduate	Training
Understanding the processes of logical resoning.	13	23.3	1
Inderstanding the role of the Enter- sion workers in predicting probable future results from existing fects.	12	12.2	2
Understanding the relationship of the "problem solving" method of effective thinking.	11	11,2	3
terateign of techniques for develop- ing effective thinking in Extension prosps.	19	10.2	4
Her to use the problem solving approach in Extension work.	9	9.2	5
Saderstanding the relation of language (turns used) to the bhicking process.	4	4.1	7
Understanding the effect of pressure groups on the thinking process.	ħ	4.1	7
knowledge of the principles of proctive thinking.	4	4.1	7
Understanding the relationship of day- dressing to orective thinking.	3	3.1	9.5
understanding the effect of psejudice on the thinking process.	3	3.1	9.5

TABLE 25.—In-corving training desired by Kanasa Agricultural Petension Agents in the general area of effective thinking.

Specific Arss of Training	Agents Desire for In- Training			
	Busber	For Gent	rent	
Moraledge of techniques for developing effective thinking in Extension groups.	28	20.6	1	
inderstanding the relationship of the problem solving method to effective which making.	19	19.4	2	
nderstanding the role of the Extension orders in predicting probable future emults from existing facts.	16	16.3	3	
dow to use the problem solving approach in Axtension work.	15	15.3	4	
Enceledge of the principles of creative taining.	12	12.2	5	
Understanding the effect of prejudice on the thinking process.	10	10.2	6	
Inderstanding the offect of pressure groups on thinking process.	7	7.1	7	
Understanding the proceess of logical rescening.	6	6.7	8	
Understanding the relationship of dep-drumning to proutive thinking.	5	5.1	9	
Understanding the relation of language (terms used) to the thinking process.	4	4.2	10	

- Knowledge of techniques for developing effective thinking in Extension groups.
- Understanding the relationship of the "problem colving" method to effective thinking.
- 3. Understanding the role of the Extension workers on predicting probably future results from existing facts.

The rank order correlation between the training neede expressed by Agricultural Extension Agents and the graduate training desired was .239. The correlation between the training needs desired and in-service training desired was .870.

There was low rank order correlation between the training needs of Agricultural Agents and the graduate training desired by Agente within the general area of "effective thinking". The correlation between the training needs of Agricultural Agente and the in-service training Agents desire to include in next years in-service training program in the district level was very high.

The data indicate that Kansae Agricultural Extension Agents could meet their expressed training needs through in-service training programs which they desire to take in the district level, to a greater degree than through the graduate training program they desire to take when they take their leaves of absence.

Program Planning Development

Table 26 shows the opinione of Kansas Agricultural Extension Agente as to their training needs within the general area of "program planning and development". The following were the three specific areas in which Agents expressed the greatest need for training within the area of "program planning

TABLE 26.—Kaness Agriculturel Extension Agents' opinions of their training needs in the general area of progrem planning and development.

Specific Area of Training	ank Order and Weighted Foores of Agents		
T.	eighted icore	Bank	
How to develop people's understanding of the county situation. (Frogram committee sembers and others).	2.01.	1.5	
Now to "build in" evaluation procedures into program plans.	2.01	1.5	
How to organize effective program planning committee.	2,00	3	
Now to involve "lay" people in program development.	1.67	4.5	
now to build an integrated county program.	1.67	4.5	
How to identify problems in the county situation and determine the priority of problems.	1.61	6	
How to analyze the situation in my county.	1.80	7	
Understanding the proper role I should assume with program planning committees.	1.75	6.	
How to use specialists in program development.	1.75	€.	
How to develop a long-time Extension program plan.	1.74	10	
How to determine objectives of the county program.	1.73	11.	
Now to relate the annual plan of work to the long- time Extension program plan.	1.73	n.	
Understanding the implications of the rapidly changing population trands in my county.	1.72	13	

and development";

- How to develop people's understanding of the county situation.
 (Program committee members and others).
- How to "build in" evaluation procedure into program plans.
 (Tied for first).
- 3. How to organize effective program planning committees.

Table 27 shows Kansas Agricultural Extension Agents' ranking of the specific areas within "progress planning and development" which they desire to include in their graduate training. The following three specific areas were the areas in which most Agricultural Extension Agents desire to study when they take their leaves of absence for advanced study:

- 1. How to organize effective program planning committees.
- 2. How to "build in" evaluation procedures into program planning.
- 3. How to involve "lay" people in program development.
- How to identify problems in the county situation and determine the priority of problems. (Tied for third).

Table 28 shows Keness Agricultural Extension Agents ranking of the specific areas within "program development" which they desire to include in their in-service training program in their district. The following are the three areas which most Agricultural Extension Agents desire to have in next year's in-service training program in their district:

- Now to develop peoples' understanding of the county situation.
 (Program committee members and others).
- 2. How to "build in" evaluation procedures into program plans.
- 3. How to organise effective program planning committees.

The following rank order correlations were computed between: (1) the training needs of Kansas Agricultural Extension Agents and the

TABLE 27.—Graduate training desired by Kansas Agricultural Extension Agents in the general area of program planning and development.

Specific area of Training	Agente Desire for Graduate Training		
	Number	Fer Cent	Mank
Now to organize effective program planning committees.	24	14.3	1
New to "build in" evaluation procedures into program plans.	13	13.3	2
How to involve "lay" people in progrem development.	11	11.2	3.5
now to identify problems in the county situation and determine the priority of			
problems.	11	11,2	3.5
How to analyze the edituation in my county.	10	10.2	5
How to develop people's understanding of the county situation. (program committee members and others).	•		-
	9	9.2	6
How to build an integrated county progress.	7	7.1	7
Understanding the proper role I should assume with program planning committees.	5	5.1	8.5
Understanding the implications of the rapidly changing population trends in my county.	5	5.1	8.5
low to determine objectives of the county program.	4	4.1	10
dos to develop a long-time Extension program	2	2.0	11.5
iow to relate the annual plan of work to the long-time Extension program plan.	2	2.0	11.5
ow to use specialists in program development.	1	1.0	13

TABLE 28.—In-service training desired by Hansas Agricultural Extension Agents in the general arm of program planning and development.

		Desire for Training	In-service Fank	
	Munber	Fer Cent		
Now to develop people's understanding of the county situation. (Program committee members and others).	18	16.4	1	
New to "build in" evaluation procedures into progress plans.	17	17.3	2	
Now to organize effective progress planning cosmittees.	16	16.3	3	
How to analyze the mituation in my county.	15	15.3	4	
Now to involve "lay" people in program development.	12	12.2	6.5	
Now to identify problems in the county cituation and determine the priority of problems.	12	12.2	6.5	
Understanding the implications of the rapidly changing population trends in my county.	12	12.2	6.5	
How to build an integrated county program.	12	12.2	6.5	
flow to relate the annual plan of work to the long-time intension program plan.	11	11,2	9	
How to determine objectives of the county program.	10	10.2	10	
New to use specialists in program isvelopment,	8	8.2	n	
ion to develop a long-time Extension program plan.	7	7.1	12	
Inderstanding the proper role I should assume with program planning committees.	4	4.1	13	

graduate training they desire to take, and (2) the training needs of Agrioultural Agents and the in-corvice training desired next year on the district level.

Training needs and graduate training desired .761
Training needs and in-service training desired .728

The rank order correlation between training needs and graduate training desired and training needs and in-service training desired by Kansas Agri-outtural Extension Agents at the district level, were moderately high and about the same for both comparisons. This correlation indicates that Kansas Agridultural Extension Agents desire graduate training, and in-service training that will tend to meet their expressed training needs in the area of "program planning and development."

Human Development

Table 29 shows the weighted scores and ranks of training needs desired by Kamens Agricultural Extension Agents within the general area of "human development". The following were the three most important specific areas in which Agricultural Agents expressed the greatest need:

- 1. How to develop my own leadership abilities.
- 2. Emplodge of the principles and techniques in effective counseling.
- 3. How to develop an approach to Entension work that considers the feelings and values of the people served by the program.

Table 30 shows the Kansas Agricultural Extension Agents' ranking of the specific areas within "human development" which they desire to include in their graduate training program. The following three specific areas were those in which most Agricultural Agents indicated they desire to include in their graduate study when they take their leaves of absence:

Tible 29. - Enous agricultural Extension Agents' spinions of their training needs in the general area of human development.

Specific Ares of Training	feore of	Bank Order and Weighted Score of Aconts		
	general	Bank		
Now to develop my own leadership abilities	2,29	1		
Enowledge of the principles and techniques in effective counseling.	2,12	2		
ion to develop an approach to intension work that considers the feelings and values of the people served by the program.	1.86	3		
inderstanding the developmental process of poys and girls.	1.85	5		
inderstanding the factors affecting the behavior of people.	1.05	5		
inowledge of the attitudes and values hald by people.	1.85	5		
inderstanding the factors affecting corsonality development in zural people.	1.62	7		
inderstanding the basic psychological drives of people, such as the need for recognition and need for security.	1.79	8		
inderstanding the reasons for aggressive selection.	1,78	9		
inderstanding why people have certain toals in life.	1.72	10		

- Enceledge of the principles and techniques in offective counseling.
- 2. Now to develop my own leadership abilities.
- 3. Understanding the factors affecting the behavior of people.

Table 31 shows the Agricultural Extension Agents' ranking of the specific eross within "human development" which they desire to include in district in-service training programs next year. The following were the most important areas which most Agricultural Agents indicated a desire for in-service training:

- 1. How to develop my own leadership abilities.
- Enculedge of the principles and techniques in effective counsuling.
- Here to develop an approach to Extension work that consider the feelings and values of the people served by the program.

The rank order correlation between the training needs desired by Eanses Agricultural Extension Agents and the graduate training they desire to include in this area was .743. The rank order correlation between the training needs desired and the in-service training Agents desired to include in their district next year was .772.

The correlations in both of the above comparisons were moderately high.

The data indicate that Kansas Agricultural Extension Agents desire inservice training and graduate training that will tend to meet their expressed
training needs in the area of "human development".

Table 30.—Graduate training desired by Kansas Agricultural Extension Agente in the general area of human development.

Specific Area of Training		Desire for Training	Graduate
	Musber	Fer Cant	Service .
Encodedge of the principles and techniques in effective counseling.	17	17.3	1
			1
Now to devolop my own loadership abilities.	16	16.3	2
Understanding the factors affecting the behavior of people.	12	12,2	3
Understanding the basic psychological drives of people, such as the need for recognition and need for recognition.	8	2.0	
seedimental and thous 191, seedinfelt	81	8.2	5
inowledge of the attitudes and values hold by papple.	8	8.2	5
lew to devulop an approach to lixtension work that consider the feelings and			
falues of the people served by the program.	8	8,2	5
Inderstanding why people have certain goals	on		
a life.	6	6.1	7.5
Inderstanding the factors affecting			
personality development in rural people.	6	6.1	7.5
mieratecting the developmental process of boys and girls.	9	5.1	9
inderstanding the ressons for aggressive			
whavior.	4	4.1	10

Tible 31. - In-service training desired by Manage gricultural Entention Agents in the general area of human development.

Specific area of Training	Agenta [†]	Desire for In-	Agents' Desire for In-service		
	mander	Fer Cent	-and		
Now to develop my own leadership abilities.	26	26.5	1		
facultodge of the principles and techniques is effective counseling.	20	20,4	2		
Now to develop an approach to Extension make that considers the feelings and values of the people served by the program.	18	18.4	3		
Inderstanding the bosic psychological frives of people, such as the need for recognition and for security.	10	30.2	4		
moviedge of the attitude and value meld by people.	9	9.2	5		
inderstanding the factors affecting the behavior of people.	8	6.2	6+5		
distributing the reseons for aggressive behavior.	8	8.2	6.5		
inderstanding the developmental process of beys and girls	6	6.1	8		
inderstanding the factors affecting person- lity development in rurel people.	5	5.1	9		
inderstanding why people have certain soals in life.	3	3.1	10		

Communications

Table 32 shows the ranks and the weighted scores of specific training needs of Kansas Agricultural Extension Agents within the general area of "communication". The following were the three specific areas in which Kansas Agricultural Extension Agents expressed the greatest need for training within the general area of "communication":

- 1. How to make my public speaking sore effective.
- 2. How to develop and use exhibits effectively.
- 3. How to write effective reports.
- 3. How to use television effectively and efficiently (tied for third).

 Table 33 shows the Kansas Agricultural Extension Agents' ranking of the specific areas within "communication" which they decire to include in their graduate study program. The following were the most important three items which Agricultural Agents desire to include in their study when they take their leaves of absence:
 - 1. How to make my public speaking nore effective.
 - 2. Now to use visual aids in teaching.
 - Knowledge of the techniques in making office calls officient and effective.

Table 34 shows the Agricultural Extension Agents' reaking of the specific areas within "communication" which they desire in next year's inservice training program in their district. The following were the most important areas which Agricultural Extension Agents desire in the district inservice training programs

Table 32.—Tenars agricultural Extension Agents' epinions of their training needs in the general area of communication.

Specific Area of Training	Rink Order and Poighted Scores of Agenta		
	Score	linak	
How to make my public speaking more effective.	2,16	1	
Her to develop and use exhibits effectively.	2,02	2	
Now to write effective reports.	1.95	3.5	
tion to use television effectively and efficiently.	1.95	3.5	
How to use photographs effectively in Extension communication.	1.92	5	
How to use visual side in terching.	1.69	6	
Now to use radio effectively and efficiently.	1.62	7	
How to write effective news articles.	1.61	8	
Now to use news letters effectively and efficiently.	1.79	9	
Encaledge of the techniques in making office dalls efficient and effective.	1.77	10	
How to use circular letters effectively.	1.75	11	
liow to write effective personal letters.	1.73	1.2	
How to conduct effective farm and home visits.	1.63	13	
Understanding the relationship between group processes and communication.	1.52	14	
Understanding the basic principles of communication.	1.37	15	

TYSUE 33.—Traducte training desired by Sancia Agricultural Priomaion Agenta in the general area of communication.

Specific Area of Training	Agents'	Decire for Training	Graduate
	250783453,	1 01, AMP	texas
Now to make my public speaking more effective.	15	15.3	1
How to use visual side in temphing.	7	7.1	2
insuledge of the techniques in making office calls efficient and effective.	5	5.1	3
now to develop and use exhibits effectively.	4	4.1	6
now to write effective personal letters.	4	4.1	6
Now to prite effective reports.	4	4.1	6
now to write offective news articles.	4	4.1	6
dow to use photographs effectively in Extension communication.	4	4.1	6
Now to use news letters effectively and efficiently.	3	3.1	9
Inderstanding the basic principles of communication.	2	2.0	1.2
Inderstanding the relationship between group processes and communication.	2	2.0	12
dus to use radio effectively and efficiently.	2	2.0	12
Tow to use circular letters effectively	2	2.0	12
Now to conduct effective farm and home visits.	2	2.0	12
How to use television effectively and efficiently.	1.	1.0	15

TABLE 34. -In-cervice training decired by Kanese Agricultural Extension Agents in the general area of communication.

How to make my public speaking more offective. Recordedge of the techniques in making office calls efficient and effective. Recordedge of the techniques in making office calls efficient and effective. Recordedge of the techniques in making office calls efficient and effective. Recordedge of the techniques in making office calls efficient and effective. Recordedge of the techniques in making office calls office ca	Specific Area of Training	Agenta Desire for In-service			
How to make my public speaking more effective. Anonhodge of the techniques in making office calls efficient and effective. Best to write effective reports. How to develop and use exhibits effectively. How to conduct effective farm and home visits. How to write effective nows erticles. How to use photographs effectively in Extension communication. How to use circular letters effectively. How to write effective personal letters. How to use radio effectively and efficiently. How to use realise effectively and efficiently. How to use news letters effectively and efficiently. Understanding the basic principles of communication. 4.1 12 How to use visual aids in teaching. How to use visual aids in teaching. How to use telegicion effectively and	· · · · · · · · · · · · · · · · · · ·			hank	
Rem to write effective reports. How to develop and use exhibits effectively. How to conduct effective farm and home visits. How to conduct effective farm and home visits. How to use photographs affectively in antension communication. How to use circular letters effectively. How to use radio effective personal letters. How to use radio effectively and efficiently. How to use radio effectively and efficiently. How to use radio effectively and efficiently. How to use news letters effectively and efficiently. Understanding the basis principles of communication. Understanding the relationship between group processes and communication. How to use visual aids in teaching. How to use talegicion effectively and	Now to make my public speaking more offective.	and the second of the second o		1.5	
How to develop and use exhibits effectively. How to conduct effective form and home visits. How to write effective news exticles. How to use photographs effectively in Extension communication. How to use circular letters effectively. 7 7.1 8 How to use circular letters effectively. 7 7.1 8 How to use radio effectively and efficiently. 6 6.1 10 How to use news letters effectively and efficiently. 6 6.1 10 Understanding the back principles of communication. 4 4.1 12 How to use visual aids in teaching. How to use visual aids in teaching. How to use telegicion effectively and		18	12.4	1.5	
Row to conduct effective farm and home visits. Row to write effective news erticles. Row to use photographs effectively in artension communication. Row to use circular letters effectively. Row to use redic effective personal letters. Row to use radio effectively and efficiently. Row to use news letters effectively. Row to use visual aids in touching. Row to use visual aids in touching. Row to use telegicion effectively and	How to write effective reports.	16	16.3	3	
How to write effective news erticles. Now to use photographs effectively in Extension communication. How to use circular letters effectively. Total 8 How to use radio effective personal letters. How to use radio effectively and efficiently. How to use news letters effectively and efficiently. How to use news letters effectively and efficiently. Understanding the basic principles of communication. Understanding the relationship between group processes and communication. How to use visual aids in teaching. How to use visual aids in teaching. How to use television effectively and	How to develop and use exhibits effectively.	15	15.3	4.5	
Now to use photographs effectively in Extension communication. How to use circular letters effectively. How to write effective personal letters. How to use radio effectively and efficiently. How to use news letters effectively and efficiently. How to use visual aids in teaching. How to use visual aids in teaching. How to use television effectively and	How to conduct effective farm and home visits.	3.5	15.3	4.5	
Extension communication. How to use circular letters effectively. Flow to write effective personal letters. How to use radio effectively and efficiently. How to use news letters effectively and efficiently. Understanding the basic principles of communication. Understanding the relationship between group processes and communication. How to use visual aids in teaching. How to use visual aids in teaching. How to use talegicion effectively and	How to write effective news exticise.	19	23,3	6.5	
How to write effective personal letters. 6 6.1 10 How to use radio effectively and efficiently. 6 6.1 10 How to use news letters effectively and efficiently. 6 6.1 10 Understanding the basis principles of communication. 4 4.1 12 Understanding the relationship between group processes and communication. 4 4.1 12 How to use visual aids in teaching. 4 4.1 12 How to use visual aids in teaching. 4 4.1 12	Now to use photographs effectively in Extension communication.	13	15.3	6+5	
How to use radio effectively and efficiently. 6 6.1 10 How to use news letters effectively and 6 6.1 10 Understanding the basic principles of 6.1 12 Understanding the relationship between group processes and communication. 4 4.1 12 How to use visual aids in teaching. 4 4.1 12 How to use visual aids in teaching. 4 4.1 12	How to use circular letters effectively.	7	7.1	8	
How to use news letters effectively and efficiently. Understanding the basic principles of communication. Understanding the relationship between group processes and communication. 4 4.1 12 How to use visual aids in tushing. 4 4.1 12	Now to write effective personal letters.	6	6.1	10	
Understanding the basis principles of communication. 4 4.1 12 Understanding the relationship between group processes and communication. 4 4.1 12 How to use visual aids in teaching. 4 4.1 12	How to use radio effectively and efficiently.	6	6.1	10	
Communication. 4 4.1 12 Understeading the relationship between group processes and communication. 4 4.1 12 How to use visual aids in tunching. 4 4.1 12 How to use telegislon effectively and	How to use name letters affectively and efficiently.	- 6	6.3	10	
group processes and communication. 4 4.1 12 How to use visual aids in touching. 4 4.1 12	Understanding the basis principles of communication.	4	4.1	12	
How to use telegicion effectively and		4	4.1	12	
	now to use visual sids in busching.	4	4.1	12	
		3	3,1	15	

- 1. How to make my public speaking more effective.
- Knowledge of the techniquee in making office calls efficient and offective. (tie for first).
- 3. How to write effective reporte.

The rank order correlation between the training neede expressed by Agricultural Extension Agents and the graduate training Agents desire to take in this area was .497. The rank order correlation between the training needs expressed by Agricultural Agente and the in-service training Agents desire next year in their district was .476.

The correlation between the expressed training needs and graduate training desired and training needs and in-service training desired at the district level, were low and about the same for both comparisons. These data indicate that Kansas Agricultural Extension Agents do not desire graduate training, and desire in-service training that will meet their expressed training needs in the area of "communication".

The Educational Process

Table 35 shows the Kansas Agricultural Extension Agente' opinions of their expressed training needs within the general area of "the educational process". The following were the three epecific areas for which Agricultural Agents expressed the greatest need for training:

- 1. Understanding how to motivate people.
- 2. How to develop and use teaching methods.
- 3. Knowledge of the principles end procedures in teaching adults.

Table 36 shows the Kansas Agricultural Extension Agents' ranking of the specific areas within "the sducational procees" which they desire to include in their graduate study program. The following three specific areas were the most important once which Agricultural Extension Agente desire to include in

TABLE 35.—Kaneas Agricultural Extension Agents' opinions of their training needs in the general area of the educational process.

Specific Area of Training	Fank Orders and Weighte Scores of Agents Weighted		
Politica II	Score	Bank	
Understanding how to motivate people.	2,22	1	
How to develop and use teaching methods.	2,00	2	
Encolodge of the principles and procedures in teaching adults.	1.98	3	
Understanding the psychological theories of learning.	1.60	4	
Understanding the philosophies of education,	1.75	5	
Understanding the effect of the differences in individuals on the teaching-learning process.	1.66	6	
Understanding the relationship between the need for and usefulness of subject matter information to learning.	1.62	7	
Understanding how people learn.	1.59	8	
Knowledge of the background and history of education.	1.58	9	
Understanding the relation of interest to learning.	1.56	10	
Understanding the relation of experience to learning.	1.51	n	
Understanding the relation of learning to what we perceive (see, feel and hear) through our senses.	1.46	12	

their graduate training program when they take their next leaves of absence:

- 1. Understanding how to motivate people.
- 2. Knowledge of the principles and procedures in teaching saults.
- 3. How to develop and use teaching methods.

Table 37 shows the Kanass Agricultural Extension Agents' ranking of the specific areas within "the educational process" which they desire-to include in their in-service training program in their district. The following were the three specific areas in which Agents indicated the greatest desire:

- 1. Understanding how to motivate people.
- 2. How to develop and use teaching methods.
- 3. Knowledge of the principles and procedures in teaching adults.

The rank order correlation between the training needs desired by Agricultural Agente and the graduate training was .821. The correlation between the training needs desired and the in-cervice training Agricultural Agente desire to include in their district training program wee .713.

The correlation between expressed training needs and graduate training desired was higher than expressed training needs and in-service training desired in the district level. The data indicate that the graduate training program would more nearly meet the expressed training needs of the Kansas Agricultural Extension Agents than would the in-service training program in the area of "the educational process".

Understanding Social Systems

Table 38 shows the opinions of Kansas Agricultural Extension Agents as to their expressed training needs within the general area of "social systems". The following were the three specific areas in which Agents expressed the greatest need for training within the area of "social systems".

TABLE 36. -- Speciate training desired by Kansan Agricultural Entension Agents in the general area of the educational process.

Specific Area of Training		Desire for C		
spectra and or manned	Musber	The same of the sa	Jank	
Understanding how to motivate people.	26	26.6	1	
Encolodge of the principles and procedures in teaching radults.	12	12.2	2	
New to develop and use teaching methods.	10	10.2	3	
Understanding how people learn.	9	9.2	4	
Understanding the psychological theories of learning.	7	7.1	5.5	
Understanding the philosophies of education.	7	7.1	5.5	
Understanding the effect of the differences is individuals on the teaching-learning process.	n. 5	5.1	7	
Understanding the relation of learning to what we perceive (see, feel and hear) through our senses.	3	3.1	8.5	
Knowledge of the background and history of education.	3	3.1	2.5	
Understanding the relation of experience to learning.	2	2.0	11	
Understanding the relatio ship between the need for and usefulness of subject matter information to learning.	2	2.0	11	
Understanding the relation of interest to learning.	2	2.0	11	

TABLE 37.—In-review training desired by Esnace Agricultural Entension Agents in the general area of the educational process.

Specific Area of Training	Agenta	Desire for In-service Training		
Made Trans. An examinal	Mumber	For Cent	vanic	
Inderstanding her to motivate people.	33	33.7	1	
low to develop and use teaching methods.	24	24.5	2	
incolledge of the principles and procedures in techning adults.	23	23.5	3	
Understanding the relationship between the need for and usefulness of subject matter information to learning.	10	10.2	Ą	
Understanding how pupple learn.	6	6.1	5.5	
Understanding the effect of the differences in individuals on the teaching-learning process	. 6	6.1	5.5	
Understanding the relation of interest to learning.	-5	5.1	- 7.5	
Understanding the philosophies of education.	5	. 5.1	7.5	
Understanding the relation of learning to what we perceive (see, feel and hear) through our senses.	4	4.1	9	
Understanding the psychological theories of learning.	3	3.1	16	
Understanding the relation of experience.	2	2.0	11	
Knowledge of the beckground and history of education.	0	0.0	12	

- 1. Understanding the effect of changing social values on family life in my country.
- 2. Understanding the interactions of individuals in groups.
- 3. Understanding why people join groups and organizations.

Table 39 shows Kanser Agricultural Extension Agents' ranking of the specific areas within "social systems" which they desire to include in their graduate study program. The following three specific areas were the most important areas which Agricultural Agents desire to include in their graduate training when they take their leaves of absence for advanced study:

- 1. Knowledge of how to identify leadership in my county.
- Understanding the effect of changing social values on family life in my county.
- Understanding the interaction of individuals in groups. (tie for third).
- 3. Understanding the relation of social systems (the family, the communication. (tie for third).

Table 40 shows the Kaness Agricultural Extension Agents' ranking of the specific areas within "social systems" which they desire to include in the in-service training programs in their district. The following were the three most important specific areas which Agents indicated they would like to include in their in-service training program:

- 1. Enowledge of how to identify leadership in my county.
- Understanding the effect of changing social values on family life in my county (tie for first).
- 4. Understanding the patterns of interdependence of the various groups in my county.

TABLE 38.—Eances Agricultural Extension Agents' opinions of their training needs in the general area of understanding social systems.

Specific Area of Troining	Penk Order and Weighted Scores of Agents		
	Score	Pent	
Understanding the effect of changing social values on family life in my county.	1.60	1	
Understanding the interactions of individuals in group.s	1.79	2	
inderstanding thy people join groups and organisations.	1.78	3	
inderstanding the relation of social systems (the family, the community, organizations and groups) to communications.	1.65	4	
Understanding the patterns of inter- dependence of the various groups in my sounty.	1.64	5	
inderstanding the functions of ferm organi- matter in rural life.	1.63	6	
inderstanding the role of the "informal" leader in the scoeptance of new practices.	1.60	7.5	
Cowledge of how to identify leadership in my county.	1.60	7.5	
inderstanding the purposes of the various public agencies in my county and their relation to the Extension Corvice.	1.55	9	
inderstanding the community organization in any county.	1.39	10	

Tible 39-Craduate training desired by Kaness Agricultural Actuation -gents in the general area of understanding social systems.

Specific Area of Training	Agenta Desire for Grad		
	MARGOS'	Fer Veris	Liter
in my county.	10	10.2	1
inderstanding the effect of changing modal values on family life in my county.	8	8.2	3
Inderstanding the intersetions of Individuals in group.s	8	8.2	3
Understanding the relation of cocial cystems (the family, the community, organizations and groups) to communication.	8	8,2	3
Indominating the purposes of the vericus sublic agencies in my county and their relation to the Extension Service.	6	6.1	5
Indorstanding the patterns of inder- inpendense of the various groups in my county.	5	5.1	6
Understanding the role of the "informal" leader in the acceptance of new practices.	4	4.1	7.5
inderstanding the functions of farm organization in rural life.	4	4.3	7.5
inderstanding why people join groups and organizations.	3	3.1	9
Understanding the community organization in any county.	1	1.0	10

Tible 40.—In-cervice training desired by manage agricultural Extension Agents in the general eres of understanding social systems.

Specific Area of Wraining	Agents	Decire for In-service Training		
charter are as a surrent	Munber	Fer Cent	Konk	
Encolodge of how to identify leadership in my county.	10	10.2	1.5	
Endorstanding the effect of changing appeal values on family life in my county.	10	10.2	1.5	
Understanding the patterns of inter- dependence of the various groups in my county.	9	9.2	4	
inderstanding the interactions of individuals in groups.	9	9.2	L _b	
Understanding the relation of social systems (the family, the community, organisations and groups) to communi- cation.	9	9.2	4	
"mierstanding the role of the "informal" leader in the acceptance of new practices.	É	6.2	6	
Under chanding why people join groups and organizations.	7	7.1	7.5	
Understanding the functions of farm proposition in turnel life.	7	7.1	7.5	
Understanding the purposes of the various public agencies in my county and their relation to the Extension	ov-			
Sarvice.	6	6.1	9	
Understanding the community organization in my county.	4	4.1	10	

- 4. Understanding the interactions of individuals in groups. (tie for fourth).
- 4. Understanding the relation of social systems (the family, the community, organizations and groups) to communication. (tie for fourth).

The renk order correlation between the training needs desired by Kansas - Agricultural Extension Agents and graduate training they desire to take in this area was .378. The rank order correlation between the training needs of Kansas Agricultural Extension Agents and the in-service training they desire to include in their district training program was .600.

The correlation between training needs and in-service training desired was higher than training needs and graduate training desired. The data indicate that the in-service training program would more nearly meet the expressed training needs of the Kansas Agricultural Extension Agents, than would the graduate training program in the area of "social systems".

Extension Organization and Administration

Table 41 shows the opinione of Kansas Agricultural Extension Agents relative to their training needs within the general area of "Extension organization and administration". The following were the most important three specific areas in which Agricultural Agents desire training within this general area:

- Understanding Extension policies and procedures on promotion and salary adjustment.
- Enouloge of the erganization and functions of the Extension Service at the Federal Level.

TABLE 41. Tensor Agricultural Entension (gents) coinfors of their tenining needs in the general area of Extension organization and administration.

Specific Area of Training	impk Unders at Stores of Wellstad	
·	geore	hank
Understanding Extension policies and procedures on promotion and salary adjustment.	1.97	3
Mnowledge of the erganization and functions of the Extension Service at the Federal level.	1.96	2
Mortedge of qualifications for Extension personnel at all levels in the Extension Service.	1.67	3
Understanding principles and procedures in office consequent.	1,61	4
Understanding how policies are formulated in the Kanses Extension Service.	2.76	545
Understanding religionship and insurance policies and procedures.	1.76	5.5
Enceledge of the existing policies in the	3.467	7
Understanding functions and responsibilities of Extension workers at the district leval.	1.65	8.
Understanding the recruitment procedures in the Intension Service in my state,	1.65	e.;
Annietge of professional improvement opportunities.	1.63	10
Understanding the functions and responsibilities of the specialist staff.	1.60	n
Understanding Extension philosophy.	1.39	12
Understanding functions and responsibilities of Extension workers at the county level.	1.29	13
Understanding the Extension History.	1.20	14
Understanding how the Hansas Extension Service is organized.	1.16	15

3. Encyledge of qualifications for Artenaton personnel at all levels in the Extension Corvict.

Table 42 shows the hances agricultural extension agents' renking of the specific area of "Extension organisation and administration" which they desire to include in their graduate training program. The following were the most important three specific areas that agricultural agents indicated they desire to include in their graduate training when they take their next leaves of absence.

- 1. Understanding principles and procedures in office management.
- 2. Understanding Extension philosophy.
- 3. Macadedge of the organization and functions of the Matenaion Service at the Fadoral level.

Table 43 shows the Europe gricultural Attention Agents' ranking of the opecific areas within "Entension organization and development" which they desire to include in their district in-convice training progress next year. The following were the most important three specific areas which Agricultural Agents indicated that they desire in in-convice training:

- 1. Understanding principles and procedures in office sepagement.
- 2. Understanding how policies are formulated in the Names Extension Corries. (tim for two).
- 2. Understanding Stitension policies and procedures on promotion and salary adjustment. (tie for two).
- 4. Understanding retimement and incurance policies and procedures.

The rank order correlation between the training needs expressed by Kansan Agricultural Entension /gents and graduate training they desire to take in this area was .591. The rank order correlation between the training needs expressed by Kansas Agricultural Agents and the in-service training

TABLE 42.—Graduate training desired by Kaneas Agriculturel Extension Agents in the general area of Extension organization and administration.

Specific Area of Training	Agents*	Desire for Graduate Training		
	humber		PARK	
Understanding principles and procedures				
in office menagement.	18	26.4	1	
Understanding Extension philosophy.	10	10.2	2	
Knowledge of the organization and functions of the Extension Service at the Federal level.	9	9.2	3	
inderstanding how policies are formulated	arry	pan		
in the Kensas Extension Service.	6	6.3	4.5	
(nowledge of the qualifications for Extension personnel at all levels in the Extension				
Bervice,	6	6.1	4.5	
inowledge of the existing policies in Hansas Extension Service.	4	4.3.	7	
inculadge of professional improvement	4	4.1.	7	
inderstanding functions and responsibilities of Extension workers at the county level.	4	4.1	7	
inderstanding Extension policies and pro- course on promotion and salary adjustment.	3	3.1	9.5	
inderstanding retirement and incurance				
olicies and procedures.	3	3.1	915	
nderstanding Extension history.	2	2.0	11.5	
nderstanding the recruitment procedures in he Extension Service in my state.	2	2.0	11.5	
nderstanding functions and responsibilities of Extension workers at the district level.	1	1.0	13.5	
nderstanding the functions and responsi- ilities of the specialist staff.	1	1.0	13.5	
nderstanding how the Kansas Extension ervice is organised.	0	0.0	15.	

TABLE 43.—In-service training desired by Kaness Agricultural Extension Agents in the general area of Extension organization and administration.

Specific Area of Training		Agents' Desire for In- Service Training Number For Cent Fank			
19	husber	Fer Cent	renk		
Understanding principles and procedures in office management.	28	20.6	1		
Inderstanding how policies are formulated in the Maneas Extension Service.	23	23.5	2.5		
Understanding Extension policies and pro- cedures on promotion and solary adjustment.	23	23.5	2,5		
Understanding retirement and insurance policies and procedures.	.22	22.4	4		
Knowledge of qualifications for Extension personnel at all levels in the Extension Service.	14	14.3	5		
Knowledge of the organization and functions of the Extension Service at the Federal Level.	13	13.3	6		
Knowledge of the existing policies in the Hansas Extension Service.	11	11.2	7.5		
Understanding the recruitment procedures in the Extension Service in my state.	11	11.2	7.5		
Knowledge of professional improvement opportunities.	9	9.2	9		
Understanding Extension philosophy.	7	7.1	10		
Understanding functions and responsibilities of Extension workers at the district level.	5	5.1	11		
Understanding the functions and responsibilities of the specialist staff.	£	4.1	12.5		
Understanding functions and responsibilities of the Extension workers at the county level.	4	4.1	12.5		
Understanding Extension history,	1	1.0	14		
Understanding how the Kanone Extension Service in organized.	0	0.0	15		

they desire to include in their district training program was .897.

The relationship between training needs expressed by Agricultural Extension Agents and graduate training desired by Agents was somewhat low in this area, while the correlation between training needs expressed by Agricultural Agents and in-service training desired was relatively high.

There data indicate that Kansas Agricultural Extension Agents will tend to meet their expressed training needs through in-service training to a higher degree than through the graduate training program.

Summary

The following were the important findings revealed by the analysis of the data presented in this chapter:

- 1. Six of the general areas of training, "agricultural subject matter", "research and evaluation", "effective thinking", "human development", "social systems", and "Extension organization and administration", showed greater correlation between the training needs expressed by the Agricultural Extension Agents and the in-service training they desire to include in their district, then between the axpressed training needs of Agricultural Agents and the graduate training they desire to take when they take their leave of absence.
- 2. Three of the general areas of training, "program planning and development", "communication", and "the educational process", had greater correlation between the training needs expressed by Agricultural Extension Agents and the graduate training Agricultural Agents desire to take, than between the expressed training needs of Agricultural Extension Agents and the in-service training Agents desire to take in the district level.

- 3. The following specific areas of training are those in which over 25 per cent of Kaness Agricultural Extension Agents indicated a desire to include in their graduate training program when they take their leaves of absence:
 - 1. Animal Science.
 - 2. Farm Management.
 - 3. Agricultural Marketing.
 - 4. Understanding How to Motivate People.
- 4. The following specific areas of training are those in which over 25 per cent of the Kansas Agricultural Extension Agents indicated a desire to include in their in-nervice training at the district level:
 - 1. Animal Science.
 - 2. Agronomy.
 - Knowledge of techniques for developing effective thinking in Extension.
 - 4. How to develop my own leadership abilities.
 - 5. Understanding how to activate people.
 - 6. Understanding principles and procedures in office management.

CHAPTER VI

SUMMARY, CONCLUSIONS AND RECOMMENDATIONS

The major purpose of this study was to analyze the training needs of Kansas County Agricultural Extension Agents as these needs were expressed by each Agent.

Specific objectives were developed to give more detail to the study. The specific objectives were:

- To describe some of the personal and situational factors relative to training of Kansas Agricultural Extension Agents. These factors included:
 - a. Tenure (experience).
 - b. Academic degrees held by Agricultural Extension Agents.
 - c. Subjects emphasized in undergraduate degrees.
 - d. Amount of graduate training already completed toward graduate degrees.
 - The perceived role of Agricultural Extension Agents, as educational or non-educational.
- 2. To determine training needs of County Agricultural Extension Agents in Kansas in relation to the nine general areas of competency identified by the National Task Force on In-Service Training.
- To identify specific items, within the nine competency areas, in which Kansas Agricultural Extension Agents felt the most need to include in graduate and in-service programs.
- 4. To determine if there were associations or relationships between training needs as expressed by the Agents and:

- a. Temare.
- b. Ferceived educational role.

This study was based largely upon the work of the National Task Force on Cooperative Extension In-Service Training. The sine areas of competency identified by the task force were:

- 1. Extension Organization and Administration.
- 2. Haman Development.
- 3. The Educational Process.
- 4. Understanding Social Systems.
- 5. Program Planning and Development.
- A. Communication.
- 7. Effective Thinking.
- E. Technical Subject Matter.
- 9. Research and Evaluation.

The author reviewed and summarized research which helped to support and substantiate stablish the situation for investigation, and to support and substantiate the findings of this particular study.

The questionneire contained 127 sub-items. The statements used within the nime areas of competency were developed by McCormick.

Each of the nine areas was divided into four parts as follows:

Part I, concerned the importance of the mine areas to the effectiveness of the Agricultural Extension Agents in Kanses. Agents could indicate the degree of importance as "absolutely essential", "highly desirable", "sould be helpful", or "not important".

Part II concerned the amount of training the Kansas Agricultural

McComalek, op. cit.

intension Agents felt they needed in the mine cross of computancy. Agents could indicate their training needs as "feel I understand", "need some training", "need much training", and "need very such training". Another item was added "not applicable to my job". A four point scale was developed and an equal distance between the four points on the scale was secured.

Part III, conserned the areas which Kansas Agricultural Extension Agents would like to include in their graduate training program when they take their leaves of absence.

Part IV, concerned the areas which Kanses Agricultural Intension Agents would like to include in their in-service training programs at the district level next year.

The date for this study were collected from 96 County Agricultural Extension Agents in Hanses by mailed questionnaires during the month of June 1962. The respondents represented 93.3 per cent of the potential. Date were pre-soded and punched on IEM cards to facilitate analysis.

A total universe and descriptive statistical techniques were used in this study. Data were analysed by masns of mean weighted ecores, percentage distributions, rank order, and excelliments of rank order correlation. The Speamen rank order correlation occilicient (Sho) was used to show the degree of relationship or association between rankings by different groups.

The conclusions and susmary of data were arranged according to the chapters.

Situational Factors Relative to Training of Kenese Agricultural Extension Agents

The moder situational factors considered and analyzed in this study were:

Tenuro. Thirty two and seven tenths per cent of Agents had less than six years tenure. Thirty per cent had over ten years service. Thirty three per cent had 6-10 years tenure, Only one Agent had less than one year of service.

Graduate Training of agents. Aighty-four Agents hold a bachelor's degree only, and 14 hold master's degrees. Sixty-two per cent of the agents had participated in graduate study.

Undergraduate Major of Agents. Thiry per cent of the Agents majored in Animal Husbandry. Over one-fourth majored in Agricultural Economics, withteen per cent majored in Agricultural Education, and 15 per cent majored in Agrenomy. Only five per cent majored in Dairy and 10 per cent majored in other fields.

Seventy three and seven tenthe per cent of the Agents received their undergraduate training at Ransas State University. Twanty per cent graduated from Oklahoma State University. The rest of the Agents graduated from the University of Missouri, University of Mebreska, other Kansas schools and out of state schools.

Job Perception as a Factor in Training: Seventy-six and five tenths
per cent of the Agents perceived their role as that of an educator. Twentythree and five tenths per cent of the Agents perceived their role as noneducators.

Relationship of Job Perception to Undergraduate Training. Sighty per cent of the Agents who majored in Agricultural Economics, 62.4 per cent of those majored in Agricultural Education, 86.7 per cent of those majored in Agronomy, 63.3 per cent of those majored in Animal Husbandry, and 80 per cent of those majored in Dairy perceived their job as educational.

Relationship of Job Perception to Graduate Trainings Minty-two and nine tenths per cent of the agents who hold master's degree perceived their role as educational. Seventy-two and eight tenths per cent of the gents who hold only a backelor's degree perceived their role as an educational one.

highly eight and nine bonths per cent of the Agents who had completed 19 or more excepter hours of graduate work perceived their role as educational, while 72.7 per cent of the Agents who had completed less than 18 hours of graduate work perceived their role as educational.

Relationship of Temmre to Job Perception. Seventy five per cent of the 0 to 5 years tenure group, 76.4 per cent of the 6 to 20 years tenure group, and 75 per cent of the 21 years and over tenure group perceived their job as educational.

Training Boods Identified by Kansas Agricultural Extension Agents in the Sine Sectoal Areas of Competency

The following survey was organized on the basis of the hypothesis established at the beginning of this study. Each hypothesis is underlined and followed by summary and conclusion.

There is no relationship between the secunt of training needed by Kniese Agricultural Extension Agents in each of the nine areas of competency and the Agents' perception of the importance of that area of competency.

Aummry of Data: Weighted scores were the basis used in ranking the nine areas of competency. Weighted scores were obtained from the Agricultural Extension Agents' indications of the degree of training needed, or from Agents' indication of the importance of those areas.

The rank order of the nine areas of competency on the basis of the training needed by Kansas Agriculturel Extension Agents was: (1) Technical Enculedge, (2) Research and Evaluation, (3) Effective Thinking, (4) Program Planning and Development, (5) Human Development, (6) Communication, (7) The Educational Process, (8) Social Systems, (9) Extension Organization and Administration.

The rank order of the nine areas of competency on the basis of the importance of those items to an Agent's effectiveness was: (1) Technical Knowledge, (2) Extension Organization and Administration, (3) Communication, (4) Program Planning and Development, (5) Research and Evaluation, (6) The Educational Process, (7) Effective Thinking, (8) Human Development, (9) Social-Systems.

The rank order correlation between the training needed by Agricultural Extension Agents and the importance of those areas to Agents' effectiveness was .216. This correlation is actual because it represents the total population.

Conclusion. The hypothesis was accepted. The findings show that there was a low positive correlation between the ranking of expressed training needs of Kansas Agricultural Extension Agents and the importance of these areas. The degree of relationship was very low. This indicates that there was no important relationship between the expressed training needs and the importance of these areas to Agents' effectiveness.

There is no relationship between the ranking of the nine general areas of competency on the basis of the amount of training needed by Kansas Agricultural Extension Agents and tenure.

Summary of date. The following rank order correlations were computed

between the approposed braining needs of Agents in the various tenure groups:

Less than one year to ten years and sleven to twenty years .504

Eleven to twenty years and twenty-one years and ever .503

Lace than one year to ten years and tuesty-one years and ever .704

These rank order correlations show that the least correlation was between the less than one year to ten years and twenty-one years and over.

The correlations between the first and second groups were considerably higher, and in the third group it was considered high.

Complesion: The hypothesis was rejected. The data suggests that training needs of Agricultural Extension Agents very slightly emong the different tenure groups, and this difference may be an important factor to densider when pleaning training programs.

There is no relationship in the ranking of the mine cross of competency of training needed by Kanens Agricultural Extension Agents compared by job priviption groups.

Summary of data: The rank order correlations between the expressed training needs of the Kanasa Agricultural Extension Agents who perceived their job as non-admessional was .734. This was a high correlation between the training needs of those two groups.

Conclusions: The hypothesis was rejected. The correlation between Agricultural Extension Agents who perceived their role as educational and those who perceived their role as non-educational was high. This indicates that whether or not an Agricultural Agent perceived his job as that of educational it had little relationship to the ranking of his training needs in the nine general areas of competency.

There is no relationship between the amount of training needed by Kanses Agricultural Entonoion Agents in each of the mine areas of compotency and the graduate training decired.

Summary of data. The rank order correlations between the agents' ranking of the nine areas of competency on the basis of the training need in those areas and the ranking of the nine areas on the basis of the graduate training agents desire to take when they take their leaves of absence was .058.

Conclusion: The hypothesis was accepted. The findings show that there was a low correlation between the ranking of training needs and the graduate training desired. The degree of relationship was very low and this indicates that there was no important relationship between training needs and graduate training desired. The implications are that Kansas agricultural futuration Agents do not desire to take graduate training in line with their expressed training needs.

There is no relationship between the amount of training needed.

By the Kanses Agricultural Extension Agents in each of the mise areas of competency and the in-relation training desired.

Summary of data: The rank order correlations between the Agents' ranking of the nine areas of competency on the basis of the training Agents felt
they needed and their ranking of the areas on the basis of the in-service
training program on the district level was 100. This descriptive statistic
was very low, and indicates that there was little correlation between the
training needs and the in-service training desired by agents on the district
level. These findings have profound implications for supervisors in planning
the in-service training programs for Agricultural Extension Agents at the
district level.

Conclusion: The hypothesis was eccepted. The findings show that there was a very low correlation between the ranking of expressed training neede of Agricultural Agents and the in-service training desired. The degree of relationship was very low which indicates that there was no important relationship between these two variables.

Specific Items in Each of the Nine Areae of Competency in which Kensas Agricultural Extension Agents Desirs Graduate and In-Service Training

This cummary includes the specific items in each of the nine areas of competency in which Kanees Agricultural Extension Agente desire greduate and in-service training. This section did not have hypotheses underlying the analysis of data. The results should be of primary interest to Administrators and Supervisors of the Kansas Extension Service. These findings may be used in other states which have organizations and programs similar to that of Kansas Extension Service.

The following were the specific areas of competency, in which at least 25 per cent of the Kansae Agricultural Extension Agents desire to include in their graduate training when they take their leaves of absence:

- 1. Animal Science.
- 2. Farm Management.
- 3. Agricultural Marketing.
- 4. Understending how to motivate people.

The following were the epecific areae of training, in which at least 25 per cent of the Kaneas Agricultural Extension Agents desire in-service training et the district level:

- 1. Animal Science.
- 2. Agronomy.

- Knowledge of techniques for developing effective thinking in Extension
- 4. How to develop my own leadership abilities.
- 5. Understanding how to motivate people.
- 6. Understanding principles and procedures in office management

Recommendations

The Kansas Extension Service in concerned with the improvement of the competency of its Agricultural Extension Agents. The following recommendations are made based on the author's interpretation of the data in this study:

- Maintain a strong organized induction training program for all new Agricultural Extension Agents and other Agents with greater emphasis on the educational role.
- Encourage a counceling system between Agricultural Extension Agents and Supervisors, in order they can give their opinions on the training needed.
- 3. Make the information presented in this study available to advisors of graduate students majoring in Extension Education, in order they may use it in counseling with students.
- 4. Make the information presented in this study available to supervisors in order they may use them in counseling with Agents.
- 5. Strengthen the present in-service training program, by placing emphasis on important items identified by Agricultural Extension Agents, but do not overlook items in the lower ranks.
- 6. Conduct similar research studies in Kansas for identifying training needs of Home Economic Agents, Club Agents, Specialists and Supervisors.

- 7. Conduct research studies similar to this, on the training needs of County Agricultural Extension Agents in other states. Compare results.
- Make a comparison between the training needs of Agricultural Extension Agents, Home Economics Agents and Club Agents in Kansas.
- Develop job descriptions for Agents in order that they may recognize more clearly their training needs.
- 10. Determine training needs of County Agricultural Extension Agents as perceived by District Supervisors. Compare these needs with those expressed by the Agents.

ACKNOWLEDGMENT

The author wishes to express his sincere appreciation to Dr. Curtis
Trent, Department of Extension, Kansas State University, for his
guidance in the execution of this work. Thanks are due also to the Kansas
Extension Service for supplying some of the materials used in collecting
data for this study. The author expresses appreciation to Dr. W. E.
Ringler and Dr. Ronald W. Campbell for serving on his graduate program
committee.

Thanks are due to all members of my family, especially to my brother Azmi Abdul-Hadi for his encouragement and financial support.

BIBLICGEAPHY

Books

- Bellows, Roger M. Psychology of Personnel in Business. 2nd Edition. Englewood Cliffs, New Jersoy: Prentice-Hall, Inc., 1954
- Blise, R. K., et al. The Spirit and Philosophy of Extension Work.
 Washington: Graduate Chool, United States Department of Agriculture
 and the Speilon Signs Phi. 1952.
- DePhillipe, Frank A., Berliner, William M. and Cribbin, James J. Management of Training Progress. Illinois: Home Wood, Richard D. Irwin, inc., 1980.
- Hogood, Margaret J. and Price, Daniel C. Statistics for Sociologists. 2nd ed. Newleed. New York: NeGraw-Hill Neet Co., Inc., 1997.
- Planty, Earl G., et al. Training Employees and Hemagers. Hew York: Ronald Press Company, 1948.
- Siegel, Sidney. Hon-paremetric Statistics for the Behavioral Science. New York: McGraw-Hill Book to., Inc., 1956.
- Yoder, Dale. Personnel Management and Industrial Eslations. 4th Edition revised. New York: Frontice-Hall, Inc., 1956.

Reports and Bulletins

- An In-service Training Program for Gooperative Extension Personnel. Seconmendations of the Mational Task Force on Gooperative Extension In-service Training, p. 29.
- Collings, Mary. "In-Service Education of Cooperative Extension Torkors."
 Federal Extension Service. "Washington: U.S. Government Printing Office, 1954.
- Induction Training for County Extension Agents. Recommendations of the Mational Task Force on Cooperative Extension In-Service Training, p. 2.
- Joint Committee Report on Extension Programs, Folicies, and Goels. U.S.D.A. Association of Land-Grant Colleges and Universities. U.S. Covernment Printing Office, Washington, D. C., August, 1940.
- Leagans, FaulyJ. Competencies Meeded by Extension Workers. Report of Conference on Extension Training. Cornell University, 1937.
- Leagane, Paul J. "Suggestions for Induction Training of County Extension Corkers", U.S.D.A., Mimocgraph.

- Miller, Paul A., et al. The Cooperative Extension Service Today—A Statement of Soope and Seasonability. Test Lensing, Michigan: Michigan State University, 1988.
- Myers, W. I. "Heeds of a Chinging Agriculture," Esport of Conference on Extension Training, Cornell, 1957. Washington: U.S.D.A. Federal Extension Sarvice.
- Percentage for Integers 1 to 399, Washington Agricultural Experiment Station Circular 341, State College of Washington: Fullmen Washington, Sept. 1958.
- Report of Fifth National Administrative Workshop. Cooperative Extension Administration, National Agricultural Attendion Center for Advanced Study, University of Misconsin, Madison, 1956, p. 53.
- U. S. Department of Agriculture, Suling Affecting Cooperative Extension Work in Agriculture and Home Economics, Bulletin So. 255 (Westington: U. N. Government Frinting Uffice, Bevised, January, 1946.

Articles and Periodicals

- Assessing and Reporting Training Needs and Progress. United States Civil Service Commission. Personnel Methods Series, No. 3, 1961.
- Bellows, Roger M., Estep, Frances M., and Scholl, Charles E., Jr. "A Tool for Analyzing Training Meeds, the Evaluation Check Lists," Personnel. American Management Association XXIX, 1953.
- Calivar, Ambrose, "Gentinuing Learning for a Changing World, " Adult Leadership, Vol. VI, No. 3, September, 1957.
- Dumber, James C. "Aurvey Training Needs," Fersonnal, Vol. 25, No. 3, 1948.
- Fryer, Bouglas H. "Training with Special Reference to its Evaluation," Personnel Psychology, TV. IV., 1951.
- McGebee, William. "Cutting Training Waste," Perconnel Psychology, I, 1948.
- The Extension Service Review, Vol. No. 12, December 1960, p. 234.
- Trent, Curtis. Forward Through Training, Cooperative Extension Service, Kansas State University, Manhatten, Kansas. May 1962, No. 5.

Unpublished Esterial

Benn, Harold W., and Aiken, Charles H. "A Look at Our Job, Factors Associated with Effective Extension Teaching in New York: A Summary of Hajor Genelusions from two Research Projects. Ithaca, New York: Cornell University, 1953.

- Cook, Benjamin D. "Comparative Analysis of Training Needs of County Agricultural Agents in Texas." Unpublished Ph. D. theels, University of Wisconsin, 1957.
- Duncan, James A. "Training Cooperative Extension Norkers—The Coordinated Approach." Cooperative Extension Service, College of Agriculture, University of Misconsin, Medison, 1957.
- Matthews, Joseph L. "A Nothed for Determining the Training Needs of County Agents as the Busis for Flanning Training Programs." Unpublished Fh.D. thesis. University of Chicago, Chicago, 1950.
- McCommick, Robert W. "An Analysis of Training Needs of County Extension Agents in Chic." Unpublished Ph.D. dissertation, National Agricultural Extension Conter for Advanced Study, University of Wisconsin, 1958.
- Frice, Mandel K. "An Analysis of In-service Training Needs Identified by County Extendion Agents in Arhancas." Unpublished Ph.D. dissertation, National Agricultural Extension Center for Advanced Study, University of Misconain, 1960.

APPINDIX

AGRICULTURE AND HOME ECONOMICS

ANSAS STATE UNIVERSITY ivision of Extension rograms and Training, Umberger Hall ANHATTAN, KANSAS

April 30, 1962

TO: COUNTY EXTENSION PERSONNEL

RE: Personnel data and training needs

Dear Co-workers:

It has became a necessity that we have a readily available personnel file on each employee of the Kansas Extension Service. We know of no way to obtain parts of this information other than by asking you.

We have delayed for some weeks sending this questionnaire to you in order that we might attach the section on "training needs." The Kansas Training Committee will use this information in determining offerings for next year's in-service training program. Also the information will be used as a part of a master's thesis here at Kansas State University.

Would you please complete the attached questionnaire (making sure all questions are answered) and return it to the Director's office by May 15.

Sincerely yours,

Wilber E. Ringler

Assistant Director

Enclosure

PERSONNEL DATA Division of Extension Kansas State University

Kansas S	tate University
IBM Column	Name
No.	Address
1. 2. 3Schedule Number	
4Deck Number	
5. Age	
1 Under 26 years	
2 26 - 35 years	
3 36 - 45 years	
4 46 - 55 years	
5 56 ~ 65 years	
6 over 65	
6. 7. 8. 9. 10. 11. When were you bor	n?
Month Day Year	
Where were you born?	
City State	
•	
12. Sex	
1 _ male	
2 female	
13. Married	
1 yes	
2 no	
14. Please check present rank at Kansa	s State University? (write in date each acquired
1 Professor	5 Assistant Instructor
2 Associate Professor	6 Graduate Assistant
3 Assistant Professor	7 Resident Assistant
4 Instructor	8 Other (specify)

15.	What is your title as stated in your current appointment?	137
	1 County Agricultural Agent	
	2 Associate or Assistant County Agricultural Agent	
	3 County Home Economics Agent	
	4 Associate or Assistant County Home Economics Agent	
	5 Club Agent	
	6 Associate or Assistant Club Agent	
	7 Specialist	
	8 District Agent	
	9 Administrator	
	10 Other (specify)	
	17. 18. What date did you enter 20. 21. Extension the first time?	
	Month Day Year	
22.	How many total years of experience have you had in Extension as of July 1 ,	1962?
	1 Less than one year	
	2 1 - 5 years	
	3 <u> 6 - 10 years</u>	
	4 11 - 15 years	
	5 16 - 20 years	
	6 21 - 25 years	
	7 <u>20 - 30 years</u>	
	8 over 30 years	
23.	What area do you serve?	
	1 County	
	2 District	
	3 Area	
	4 State	
24.	Where do you headquarter or where is your office located?	
	1 In the county 2 Out in the district 3 At State Office in Manh	attan

25.	What is the highest degree you now	hold?
	1 Bachelor's	
	2 Master's	
	3 Doctor's	
	4 Other (specify)	
26.	Where did you receive your Bachelor	's degree?
	1 Kansas State University	
	2 Other Kansas School, (specify)	
	3 _ Oklahoma State University	
	4 University of Missouri	
	5 Colorado State University	
	6 University of Nebraska	
	7 _ Other out of state school, (sp	ecify)
	When did you receive your Bachelor'	s degree?
	Month Day Year	
27.	What was your undergraduate major?	
	1 Agriculture Economics	
	2 Agriculture or Home Economics	Education
	3 Agronomy	
	4 Animal Husbandry	
	5 Dairy	
	6 Foods and Nutrition	
	7 Clothing	
	8 Home Furnishings	
	9 Other (specify)	
28.	What was your undergraduate grade a	verage?
	1 A	5 <u> </u>
	² — B+	6 <u> </u>
	3 B	7 D

29	. What was your grade average for your last four semesters of undergraduate work?
	1 A
	2 B+
	3 <u> </u>
	4 C+
	5 <u> </u>
	6 D+
	7 <u> </u>
30 .	Number of semester hours of graduate work completed?
	0 None
	1 1 - 6
	2 7 - 12
	3 13 - 18
	4 19 - 24
	5 24 - 30
	6 Over 30
31.	Do you have an advanced degree program in progress at the present time?
	1 yes
	2 no
	If yes, specify whether Master's or Ph. D.
	Number of hours toward the degree?
	Name of school in which you are enrolled?
32.	If you hold an advanced degree, what was your field of specialization in work for the highest degree you hold?
	1 Education
	2 Agriculture or home economics education or Extension education
	3 Administration (public, education or Extension)
	4 Biological Sciences
	5 Social Sciences
	6 Home Economics
	7 Other (specify)

2 Other (specify)	
34. 35. 36. When did you receive your ac 37. 38. 39.	lvanced degree? (the highest degree you now hold)
Month Day Year	
40. Have you ever used sabbatical leave?	
1 yes	
2 no	
If yes, when?	_
41. Which of the following years did you	attend Regional Extension School?
School Courses Taken	redits Grade Type of Financial Assistance
1 1958	
2 1959	
3 1960	
4 1961	
5 1962	
42. Scholarships you have received? Date	e Received?
1 Horace Moses	
2 Farm Foundation	
3 Sears Foundation	
4 Kansas State University Subsista	ance Grant
5 Other (specify)	
43. Fellowships you have received? (write	e date received after fellowship checked)
1 Frysinger	Ford Foundation
2 Pfizer	Kellogg Foundation
3 Hall	B Farm Foundation
4 Tyson	OOther (specify)
5 USS	

44.	Have you completed the five weeks Induction Training Program inaugurated at Kansas State University in July 1958?
	1 yes
	2 no
	If yes, date of completion
45.	How many years did you live on a farm after age 10 and previous to entering Extension work? (Include years in college if you returned to farm during the summer)
	1 None
	2 _ 1 - 6
	3 7 - 12
	4 13 - 18
	5 19 - 24
	6 Over 24
46.	How many years were you a 4-H Club member?
	1 None
	2 _ 1 - 3
	3 4 - 6
	4 7 - 9
	5 Over 9
47.	Membership in an organization
	1 Lions
	2 Kiwanis
	3 Rotary
	4 Altrusa
	5 Optomist
	6 Chamber of Commerce
	7 Business & Professional Women's Club
	8 Other (specify)
48.	Membership in Learned Societies
	1 yes

	142
49.	If a county worker, in which district do you work?
	1 Northeast
	2 Southeast
	3 Central
	4 Northwest
	5 Southwest
50.	FOR COUNTY WORKERS ONLY
	Role perception by the agent himself (check the <u>one</u> which you feel most nearly describes your job)
	1 A professional agriculturist, home economist or youth worker available to provide information for the people of your county.
	A professional agriculturist, home economist or youth worker providing service to the people of your county.
	A professional educator developing educational programs with people to effect behavior changes in the people of your county.
	4 A professional educator developing programs to help people help themselves.
	5 _ A professional organizer of educational activities for the people of your county.

General Instructions

- Please respond to ALL items listed in the questionnaire.
 Do not give any attention to the deck or column numbers.
- The questionnairs are numbered for research control purposes only.
- No names of any of the respondents will appear in thesis form nor in any material published as a result of this study.
- 4. On every page you will find a list of items, please check only one column at right for each item. The checking should be based on your feeling of the importance, and need of those items to you.
- Your responses to all questions in this questionnaire should be made in terms of your most sincere feelings and beliefs.
- Please do not feel that there is a certain way that you are expected to answer.
- 7. The data collected in this study will be analyzed and the information obtained will be made available to the staff of the Kansas Extension Service to be used in further development of its graduate and in-service training program.
- Considering your present job, please indicate in the column at the right, the amount of training you feel you need in each of these areas.

4.0	EXTENSION ORGANIZATION AND ADMINISTRATION	SOF 2	TING				144	
	(Check (V) only one column at right for cach item)	NOT APPLICABL: TO MY	NEED VERY MUCH TRAINE.	TEED MUCH TRAINING	NEED SOME TRAINING	GIBEL I UNDERSTAND	Drek	
1.	Understanding Extension History.						51	
2.	Understanding Extension philosophy.						52	
3.	Understanding how the Kansas Extension Service is organized.						53	
4.	Understanding how policies are formulated in the Kanaaa Extension Service.						54	
5.	Knowledge of the existing policies in the Kansas Extension Service.						55	
6.	Knowledge of professional improvement opportunities.						56	
7.	Understanding functions and responsibilities of extension workera at the county level.						57	
8.	Understanding functions and responsibilities of extension workers at the district level.						58	
9.	Understanding the functions and responsibilities of the specialist staff.			1			59	
10.	Knowledge of the organization and functions of the Extension Service at the Federal level.						60	
11.	Understanding principles and procedurea in office management.						61	
12.	Understanding the recruitment procedures in the Extension Service in my state.						62	
13.	Understanding Extension policies and procedures on promotion and salary adjustment.						63	
14.	Understanding retirement and insurance policies and procedures.						64	
15.	Knowledge of qualifications for extension personnel at all levels in the Extension Service.						65	
When I take my next leave of absence for advanced study from the Extension Service I would like to include: None of the above areas in my graduate study program. Of the above areas in my graduate program. (no more than 3 total) (66-71) I would recommend for next year's in-service training program in my district that there be included: None of the above areas. Number(s) of the above areas.								

	PROGRAM PLANNING AND DEVELOPMENT						145
	(Check (√) only one column at right for each item.)	NOT APPLICABLE TO MY JOB	NEED VERY MUCH TRAINING	NEED MUCH TRAINING	NEED SOME TRAINING	FEEL I UNDERSTAND	Deck 2
1.	How to analyze the situation in my county.		ļ				51
2.	How to organize effective program planning committees.						52
3.	Understanding the proper role I should assume with program planning committees.						53
4.	How to involve "lay" people in program development.						54
5.	How to develop people's understanding of the county situation. (Program committee members and others)						55
6.	How to identify problems in the county situation and determine the priority of problems.						56
7.	Understanding the implications of the rapidly changing population trends in my county.						57
8.	How to develop a long-time Extension program plan.						58
9.	How to determine objectives of the county program.						59
0.	How to build an integrated county program.						60
1.	How to relate the annual plan of work to the long- time Extension program plan.						61
2.	How to "build in" evaluation procedures into program plans.						62
3.	How to use specialists in program development.						63

None of the above areas in my graduate study program. Number(s) of the above areas in my graduate program. (No more than 3 total) (84-69)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number(s) of the above areas.

(No more than 3 total) (70-75)

							10
	UNDERSTANDING SOCIAL SYSTEMS		11	ı	1	,	146
	(Check (🗸 <u>only one</u> column at right for each item)	NOT APPLICABLE TO MY JOB	NEED VERY MUCH TRAINING	NEED MUCH TRAINING	NEED SOME TRAINING	FEEL I UNDERSTAND	Deck
1.	Understanding the community organization in my county.	11					51
2.	Understanding the patterns of interdependence of the various groups in my county.						52
3.	Understanding the role of the "informal" leader in the acceptance of new practices.						53
4.	Knowledge of how to identify leadership ih my county.						54
5.	Understanding the effect of changing social values on family life in my county.						55
6.	Understanding why people join groups and organizations.						56
7.	Understanding the functions of farm organization in rural life.						57
8.	Understanding the interactions of individuals in groups.				1		58
9.	Understanding the purposes of the various public agencies in my county and their relation to the Extension Service.						59
10.	Understanding the relation of social systems (the family, the community, organizations and groups) to communications.						60
		1-11	2 1	3	ч	5	

None of the above areas in my graduate study program. of the above areas in my graduate program.

1) (61-66) (No more than 3 total)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number (s) of the above areas.

(No more than 3 total) (67-72)

	HUMAN DEVELOPMENT						
	(Check (🗸 only one column at right for each item)	NOT APPLICABLE TO MY JOB	NEED VERY MUCH TRAINING	REED MUCH TRAINING	NEED SOME TRAINING	FEEL I UNDERSTAND	Deck
1.	Understanding the developmental process of boys and girls						51
2.	How to develop my own leadership abilities.						52
3.	Understanding the basic psychological drives of people, such as the need for recognition and need for security.						53
4.	Understanding the factors affecting the behavior of people.						54
5.	Knowledge of the attitudes and values held by people.						55
6.	Understanding why people have certain goals in life.		:				56
7.	Understanding the reasons for aggressive behavior.					-	57
8.	Knowledge of the principles and techniques in effective counseling.						58
9.	How to develop an approach to Extension work that considers the feelings and values of the people served by the program.						59
10.	Understanding the factors affecting personality development in rural people.						60
		1	2	3	4	5	

None of the above areas in my graduate study program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (61-66)

I would recommend for next year's in-service training program in my district that there be includeed:

Number(s) of the above areas.

(No more than 3 total) (67-72)

							12
	THE EDUCATIONAL PROCESS	ſ 1			1	3	48
	(Check (V) only one column at right for each item)	NOT APPLICABLE TO MY JOB	HEED VERY MUCH TRAINING	MEED MUCH TRAINING	HEED SOME TRAINING	FEEL I UNDERSTAND	Deck COI
1.	Understanding how people learn.						51
2.	Understanding the effect of the differences in individuals on the teaching-learning process.						52
3.	Understanding the relation of experience to learning.						53
4.	Understanding the relation of learning to what we perceive (see, feel and hear) through our senses.						54
5.	Understanding the relationship between the need for and usefulness of subject matter information to learning.						55
6.	Understanding the relation of interest to learning.						56
7.	Knowledge of the principles and procedures in teaching adults.						57
8.	How to develop and use teaching methods.			_			58
9.	Knowledge of the background and history of education.						59
10.	Understanding the psychological theories of learning.		<u></u>	_	_		60
11.	Understanding the philosophies of education.	1		_			61
12.	Understanding how to motivate people	1					62

None of the above areas in my graduate program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (63 - 68)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number(s) of the above areas.

(No more than 3 total) (69-74)

	COMMUNICATION	Z JOB	NING	Į.	i	14	49
	(Check (🗹 only one column at right for each item)	HOT APPLICABLE TO MY	HEED VERY MUCH TRAINING	NEED MUCH TRAINING	NEED SOME TRAINING	EEL	Drc
1.	Understanding the basic principles of communication.						51
2.	Understanding the relationship between group processes and communication.						52
3.	How to make my public speaking more effective.			į			53
4.	How to develop and use exhibits effectively.						54
5.	How to write effective personal letters.	Ī				:	55
6.	How to write effective reports.						56
7.	How to write effective news articles.					:	57
8.	How to use radio effectively and efficiently.					!	58
9.	How to use television effectively and efficiently.					- :	59
10.	How to use visual aids in teaching.						60
11.	How to use news letters effectively and efficiently.	!				- 1	61
12.	How to use photographs effectively in Extension communication.					1	62
13.	How to use circular letters effectively.				Ì		63
14.	How to conduct effective farm and home visits.					- 0	64
15.	Knowledge of the techniques in making office calls efficient and effective.						65
		1	2	3	11	5	

None of the above areas in my graduate study program. Number(s) of the above areas in my graduate program. (No more than 3 total) (66-71)

I would recommend for next year's in-service training program in $my\ district\ that$ there be included:

Number(s) of the above areas.

(No more than 3 total) (72-77)

	TECHNICAL KNOWLEDGE					150	14
MALE	(Check (V) only one column at right for each item) PERSONNEL ONLY CHECK THIS PAGE	NOT APPLICABLE TO MY JOB	NEED VERY MUCH TRAINING	NEED MUCH TRAINING	NEED SOME TRAINING	FEEL I UNDERSTAND	Deci Col
1.	Animal Science.						51
2.	Dairy Science.						52
3.	Agronomy (include crops, soils, fertilizers)						53
4.	Agricultural Marketing. (include livestock, grain, dairy and poultry)						. 54
5.	Farm Management.						55
6.	Dairy Technology.						56
7.	Agricultural Engineering. (include machinery, buildings and drainage)						57
8.	Entomology.						58
9.	Horticulture.(include pomology and vegetable crops)						59
ï.O .	Soil Conservation.						60
11.	Poultry.						61
12.	Agricultural Credit.						62
13.	Consumer Marketing. (Market information for consumers)						63
14.	Floriculture.						64
15.	Plant Pathology.						65

None of the above areas in my graduate training program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (66-7/)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number(s) of the above areas (No more than 3 total) (72-77)

						15	1 15
	TECHNICAL KNOWLEDGE (Check (only one column at right for each item)	SLE TO MY JOB	MUCH TRAINING	TRAINING	TRAINING	UNDERSTAND	
4.	LE PERSONNEL ONLY CHECK THIS PAGE	NOT APPLICABLE	NEED VERY MI	NEED MUCH TR	NEED SOME TR	FEEL I UNDER	Deck
1.	Clothing Economics.						51
2.	Clothing Construction.						52
3.	New Developments in Textiles.						53
4.	Dress Design.						54
5.	Basic nutritional principles.						55
6.	Family Relations.						56
7.	Child Nutrition.						57
8.	Family Economics.						58
9.	Home Management.						59
10.	Child Development.						60
11.	Consumer Marketing (marketing information for consumers)					-	61
12.	Home Furnishings.						62
13.	Household Equipment.						63
14.	Interior Design.						64
15.	Housing.						65

None of the above areas in my graduate training program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (66-71)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number(s) of the above areas.

(No more than 3 total) (72-77)

							- 4
	EFFECTIVE THINKING	1	11	i	1	ı	
	(Check (only one column at right for each item)	NOT APPLICABLE TO MY JOB	NEED VERY MUCH THAINING	NEED MUCH TRAINING	NEED SOME THAINING	FEEL I UNDERSTAND	Dec
1.	Understanding the processes of logical reasoning.						51
2.	Understanding the relation of language (terms used) to the thinking process.						52
3.	Understanding the relationship of day-dreaming to creative thinking.						53
4.	Understanding the relationship of the "problem solving" method to effective thinking.						54
5.	Understanding the effect of prejudice on the thinking process.						55
6.	Understanding the effect of pressure groups on the the thinking process.						56
7.	Knowledge of techniques for developing effective thinking in Extension groups.						57
8.	Knowledge of the principles of creative thinking.						58
9.	How to use the problem solving approach in Extension work.						59
0.	Understanding the role of the Extension workers in predicting probable future results from existing facts.						60

None of the above areas in my graduate training program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (61-66)

I would recommend for next year's in-service training program in my district that there be included:

Number(s) of the above areas.

(No more than 3 total) (67-72)

						153	17
	RESEARCH (Check (r) only one column at right for each item)	NOT APPLICABLE TO MY JOB	NEED VERY MUCH TRAINING	NEED MUCH TRAIMING	NEED SOME TRAIMING	FEEL I UNDERSTAND	Deck Cal.
1.	Knowledge of the methods for measuring results of programs.						51
2.	How to use the experimental approach (pilot projects) in extension work.						52
3.	How to interpret the results of research in the physical sciences.						53
4.	How to interpret the results of research in the accial aciences.						54
5.	Understanding research terminology.						55
6.	Making application of research findings in assisting people in my county.				l		56
7.	Understanding ways of designing evaluation projects.						57
8.	Understanding how to interpret statistical results.						58
9.	How to develop simple survey devicea.						59
10.	Understanding the role of investigation in solving problems with people.						60
11.	Knowledge of where to secure sound research information other than from experiment stations.						61
12.	How to make the best use of research publications.		2	2		_	62

None of the above areas in my graduate study program.

Number(s) of the above areas in my graduate program.

(No more than 3 total) (63-69)

I would recommend for next year's in-service training program in my district that there be included:

None of the above areas.

Number(s) of the above areas.
(No more than 3 total) (69-74)

On the preceding pages you have had an opportunity to indicate your feeling of need for training in a large number of areas. Perhaps there are some additional areas that were not listed in which you feel a need for training. If this is the case, please indicate the area or areas below by writing them in the appropriate space and checking at the right the degree of training you feel that you need.

If you do not feel the need for training in additional areas, write NONE in the space below.

Additional Area(s) of Training	Very Much Training Needed	Much Training Needed	Some Training Necded
,			

Bclow is a suggested list of categories in which it might be desirable for county Extension agents to be trained. Please check these nine categories on the scale at the right below based on your feeling of the importance of these areas in contributing to the effectiveness of a county Extension agent.

Category	4 Absolutely Essential	3 Highly Desirable	2 Would be Helpful	1 Not Important
Extension Organization and				
luman <u>Development</u> . i.e., devel- pmental processes of people, group interaction principles.				
The Educational Process: principles of learning; teaching methods and philosophy of education				
Social Systems: family, commun- ty, school, church groups, special interest groups, farm organizations.				
Program Planning and Develop- ment.				
Communication: basic communication, individual, group and mass media.				
Effective Thinking: problem solving method; decision making based on critical analysis; and creativity.				
Technical Knowledge: subject matter in agriculture and home economics.				
<u>Research</u> : principles of research and cvaluation; methods of utilizing research findings.				

Listed below is the same list of categories you have just classified as to their importance in contributing to the effectiveness of a county Extension agent.

Please indicate by checking the scale below the amount of training you feel you need in each of these nine arcas.

Category	NOT APPLICABLE TO MY JOB	HEED VERY MUCH TRAINING	NEED MUCH TRAINING	NEED SOME TRAINING	FEEL I UNDERSTAND	Derk Col
Extension Organization and Administration						60
<u>lluman Development</u> : i.e., developmental processes of people, group interaction principles.						61
The Educational Process: principles of learning; teaching methods and philosophy of education.						62
Social Systems: family, community, school, church groups, special interest groups, farm organizations						63
Program Planning and Development						64
Communication: basic communication, individual, group and mass media.						65
Effective Thinking: problem solving methods, decision making based on critical analysis, creativity.						66
Technical Knowledge: subject matter in agriculture and home economic.						67
Research: principles of research and evaluation; methods of utilizing research findings.						68
	1	2	3	4	5	

Please check back through the entire questionnaire to be sure that all items are checked - Thank you!

AN ANALYSIS OF THE EXPRESSED TRAINING NEEDS OF COUNTY AGRICULTURAL EXTENSION AGENTS IN KANSAS

by

NAZEEH ABDUL-HADI

B.S., Kansas State University Manhattan, Kansas, 1961

> AN ABSTRACT OF A WASTER'S THESIS

submitted in partial fulfillment of the

requirements for the degree

MASTER OF SCIENCE

Department of Education

KANSAS STATE UNIVERSITY Manhattan, Kansas The major purpose of this study was to describe and analysis the expressed training needs of County Agricultural Extension Agents in Kansas. Specific objectives were developed to serve as guides in the callection and analysis of the data. The specific objectives established for the study were: (1) To describe some of the personal and situational factors relative to training of Kansas Agricultural Extension Agents. These factors were: Tenure, Academic degrees held by Agents, subjects emphasized in undergreduate degrees, amount of graduate training already completed toward graduate degrees and the perceived role of Agents, as admentional or non-educational.

(2) To determine training meeds of Agents in relation to the nine general areas of competency identified by the National Tank Force on In-Service Training.

(3) To identify specific items, within the nine competency areas, in which Kanens Agents felt the most need to include in graduate and in-service training programs. (4) To determine if there were associations or relationships between training needs as expressed by Agents and: Tenure, perceived aducational role.

The data were collected from 96 Agents by mailed questionnaire during June, 1962. A total universe and descriptive statistical techniques were used.

Data were pre-ceded and punched on IMM cards to facilitate analysis. Data were analysed by means of mean weighted scores, percentage distribution, and coefficient rank correlation (Roo).

Hime areas of competency, identified by the Matienal Task Force on In-Service Training, formed the framework for this study. The areas are. (1) Extension Organization and Administration, (2) Human Development, (3) The Educational Process, (4) Understanding Social Systems, (5) Program Planning and Development, (6) Communication, (7) Effective Thinking, (6) Technical Subject Matter, (9) Research and Evaluation.

Some of the findings revealed in the study were:

- The Kansus Literalan Service selects its Agents from the undergraduate areas of Agriculture.
- Fourteen per cent of Kaneas Agricultural Extension Agents hold master's degrees.
- 3. Thirty three per cent of the Agricultural Agents had 6-10 years temmes.
- 4. The highest per cont of Agricultural Agents majored in Amimal Nucbendry.
- Seventy three and seven tenths per cent of the Agents received their undergraduate training at Maners State University.
- Sixty two per cent of the Agents have participated in graduate training.
- Seventy six and five tents per cent of the Agents perceived their role as that of an educator.
- 8. Minty two and mine tenths per cent of Agents who held master's degrees perceived their role as educational. This compared with per cent of these holding only bachelor's degrees who perceived their role as educational.
- Agents with five to twenty years of tenure perceived their role as educational to a greater extent than Agents in other tenure groups.
- 10. The rank order of the nine areas of competency on the basis of training Kansas Agricultural Agents felt they needed wast (1) Technical Knowledge, (2) Seesarch and Evaluation, (3) Affective Thinking, (4) Program Flamming and Development, (5) Human Development, (6) Communication, (7) The Educational Process, (8) Social Systems, (9) Extension Organization and Administration.
- 11. There was no important relationship between the ranking of the expressed training needs and the ranking of importance of these areas to Agents' effectiveness.
- 12. Training needs of Agents vary alightly among the different temure groups.
- 13. Whether or not Agents perceived their role as educational showed little relationship to the ranking of training needs in the general areas of training needs.
- 14. There was no important relationship between expressed training needs and graduate training desired.

- There was no important relatio ship between expressed training needs and in-service training desired.
- 16. The following were the specific areas, which at least 25 per cent of Agents decired to include in their graduate training: (1) Animal Science, (2) Farm Management, (3) Agricultural Marketing, (4) and Understanding New to Motivate Feeple.
- 17. The following were the specific areas, which at least 25 per cent of Agents desired to include in their in-service training program:
 (1) Animal Science, (2) Agreemy, (3) Knowledge and Techniques for Developing Effective Thinking in Extension, (4) How to Develop My Own Leadership Abilities, and (5) Understanding Principles and Procedures in Office Management.

The findings of this study have their first implication to the Kanses
Extension Service. The findings should be given consideration when planning
future training programs.

The following recommendations were made based on the review of the literature and interpretation of the data analyzed in this study: (1) Maintain strong organized induction and in-corvice training programs for all Agents, with greater emphasis on the educational role, (2) encourage a counseling system between Agents and Supervisors, in order they may express their epinions as to the training needed. (3) Make information presented in this study available to student edvisors and Supervisors for use in counseling with students and Agents. (4) Strengthen the present in-service training program, by placing caphagis on important items identified by agents, but do not over look items in the lower ranks. (5) Conduct similar research studies in Kansas and other states to identify training needs of Agricultural Agents, Home Economics Agents, and Club Agents, then compare results. (6) Develop job descriptions for Agents in order that they may recognize more clearly their training needs. (?) Determine training needs of County Agricultural Extension Agents as perceived by district supervisors. Compare these needs with those expressed by the Agents.